Housing, Land, and Property Restitution Rights of Refugees and Displaced Persons

Laws, Cases, and Materials

Edited by Scott Leckie

CAMBRIDGE www.cambridge.org/9780521858755

This page intentionally left blank

HOUSING, LAND, AND PROPERTY RESTITUTION RIGHTS OF REFUGEES AND DISPLACED PERSONS

The legal recognition of the housing, land, and property rights of refugees and displaced persons has expanded steadily in recent years as the realization has grown that securing these rights will be beneficial to long-term peace, stability, economic vitality, and justice.

This volume contains more than 240 of the laws, cases, and materials that have been adopted during the past century that accord those unjustly and arbitrarily displaced from their homes and lands with rights: not simply to return to their countries or places of origin, but to return to the original home, land, or property from which they were initially forced to flee.

The breadth of the restitution standards found within this volume, combined with selected examples of case law and other materials, is a clear indication that a right to housing, land, and property restitution for refugees and displaced persons has emerged within the global legal domain.

Scott Leckie is an international human rights lawyer and advocate. He is active in many fields, including economic and social rights, housing rights, forced evictions, housing and property restitution rights for refugees and displaced persons, human rights issues in postconflict and postdisaster situations, and business and human rights.

ALSO BY SCOTT LECKIE

United Nations Peace Operations and Housing, Land and Property Rights: Proposals for Reform (ed., 2007)

Returning Home: Housing and Property Restitution Rights of Refugees and Internally Displaced Persons, Volume 2 (ed., 2007)

Legal Resource Guide on Economic, Social and Cultural Rights (ed., with Anne Gallagher, 2006)

Returning Home: Housing and Property Restitution Rights of Refugees and Internally Displaced Persons (ed., 2003)

National Perspectives on Housing Rights (ed., 2003)

When Push Comes to Shove: Forced Evictions and Human Rights (1995)

Destruction by Design: Housing Rights Violations in Tibet (1994)

From Housing Needs to Housing Rights (1992)

Housing, Land, and Property Restitution Rights of Refugees and Displaced Persons

Laws, Cases, and Materials

Edited by

SCOTT LECKIE

Executive Director Centre on Housing Rights and Evictions (COHRE)



CAMBRIDGE UNIVERSITY PRESS Cambridge, New York, Melbourne, Madrid, Cape Town, Singapore, São Paulo

Cambridge University Press The Edinburgh Building, Cambridge CB2 8RU, UK Published in the United States of America by Cambridge University Press, New York www.cambridge.org Information on this title: www.cambridge.org/9780521858755

© Cambridge University Press 2007

This publication is in copyright. Subject to statutory exception and to the provision of relevant collective licensing agreements, no reproduction of any part may take place without the written permission of Cambridge University Press.

First published in print format 2007

ISBN-13 978-0-511-28497-7 eBook (Adobe Reader) ISBN-10 0-511-28259-1 eBook (Adobe Reader) ISBN-13 978-0-521-85875-5 hardback ISBN-10 0-521-85875-5 hardback

Cambridge University Press has no responsibility for the persistence or accuracy of urls for external or third-party internet websites referred to in this publication, and does not guarantee that any content on such websites is, or will remain, accurate or appropriate. For all of you ... Blameless Yet punished, condemned To everyone but you ... nameless Gazing Eyes of hope, worn by fear And yet Longingly, The edges of a smile lurks Dreams of home Perpetually near

Acknowledgments			page xvii	
Prej	face		xix	
PAR	T ONE	. HOUSING AND PROPERTY RESTITUTION STANDARDS – INTERNATIONAL	1	
1	Inter	national Standards	1	
	Intern	national Humanitarian Law	1	
	1.1 1.2	Hague Convention (IV) Respecting the Laws and Customs of War on Land (1907) Geneva Convention (IV) Relative to the Protection of Civilian Persons in Time	1	
		of War (1949)	1	
	1.3	Protocol Additional to the Geneva Conventions of 12 August 1949, and Relating to		
		the Protection of Victims of International Armed Conflicts (Protocol 1) (1977)	2	
	1.4	Protocol Additional to the Geneva Conventions of 12 August 1949, and Relating to		
		the Protection of Victims of Non-International Armed Conflicts (Protocol II) (1977)	3	
	Intern	national Human Rights Law	3	
	1.5	Universal Declaration of Human Rights (1948)	3	
	1.6	International Convention on the Elimination of All Forms of Racial Discrimination		
		(1965)	4	
	1.7	International Covenant on Economic, Social and Cultural Rights (1966)	4	
	1.8	International Covenant on Civil and Political Rights (1966)	4	
	1.9	Convention on the Rights of the Child (1989)	5	
	1.10	Convention (No. 169) Concerning Indigenous and Tribal Peoples in Independent		
		Countries (1989)	6	
	Intern	national Criminal Law	6	
	1.11	Rome Statute of the International Criminal Court (1998)	6	
	Intern	national Refugee Law	11	
	1.12	Convention Relating to the Status of Refugees (1951)	11	
	1.13	Protocol Relating to the Status of Refugees (1967)	11	
	The I	nternational Law of State Responsibility	12	
	1.14	International Law Commission's Draft Articles on Responsibility of States for		
		Internationally Wrongful Acts (2001)	12	
	Speci	fic Restitution Standards	13	
		'Pinheiro' Principles on Housing and Property Restitution for Refugees and		
		Displaced Persons (2005)	13	
	1.16	Basic Principles and Guidelines on the Right to a Remedy and Reparation for		
		Victims of Violations of International Human Rights and Humanitarian Law (2005)	20	

2	Peace	e Agreements	25
-	2.1	Cambodia – Agreement on a Comprehensive Political Settlement on the Cambodia	
	2.1	Conflict (1991)	25
	2.2	El Salvador – Peace Agreement between the Government of El Salvador and the	20
	2.2	Frente Farabundo Martí para la Liberación Nacional (1992)	28
	2.3	Mozambique – The Rome Process: General Peace Agreement for Mozambique (1992)	30
	2.4	Liberia – Cotonou Agreement (1993)	30
	2.5	Rwanda – Arusha Peace Agreement (1993)	31
	2.6	Bosnia-Herzegovina – General Framework Agreement on Peace in	51
	2.0	Bosnia-Herzegovina (Dayton Peace Agreement) – Annex 7: Agreement on Refugees	
		and Displaced Persons (1995)	31
	2.7	Guatemala – Agreement on Identity and Rights of Indigenous Peoples (1995)	34
	2.8	Croatia – The Erdut Agreement (1995)	36
	2.9	Tajikistan General Agreement on the Establishment of Peace and National	00
	2.0	Accord – Protocol on Refugee Issues (1997)	37
	2.10	Sierra Leone – Lomé Peace Agreement (1999)	38
	2.11	Burundi – Arusha Peace and Reconciliation Agreement for Burundi (2000)	38
	2.12	Ethiopia and Eritrea – Agreement between the Government of the Federal	00
	2.12	Democratic Republic of Ethiopia and the Government of the State of Eritrea (2000)	41
	2.13	Liberia – Comprehensive Peace Agreement between the Government of Liberia and	11
	2.10	the Liberians United for Reconciliation and Democracy (LURD) and the Movement	
		for Democracy in Liberia (MODEL) and Political Parties (2003)	43
	2.14	Sudan – Comprehensive Peace Agreement – Agreement on Wealth Sharing during	10
		the Pre-Interim and Interim Period (2004)	43
3	Volui	ntary Repatriation Agreements	
0	3.1	Socialist Republic of Vietnam and UNHCR (1988)	45
	3.2	Republic of South Africa and UNHCR (1991)	43 46
	3.3	Guatemala and UNHCR (1991)	40
	3.4	United Republic of Tanzania, Democratic Republic of the Congo and UNHCR (1991)	40
	3.5	Islamic State of Afghanistan, Republic of Iran and UNHCR (1992)	47
	3.6	Republic of Mozambique, Zimbabwe and UNHCR (1993)	47
	3.7	Union of Myanmar and UNHCR (1993)	47
	3.8	Abkhazia, Georgia, Russian Federation and UNHCR (1994)	48
	3.9	United Republic of Tanzania, Republic of Rwanda and UNHCR (1995)	51
	3.10	Republic of Angola and UNHCR (1995)	51
	3.11	Republic of Liberia and UNHCR (1996)	52
	3.12	Republic of Burundi, United Republic of Tanzania and UNHCR (2001)	52
	3.13		02
	0110	and UNHCR (2002)	53
	3.14	United Kingdom of Great Britain and Northern Ireland, the Transitional	00
	0111	Administration of the Transitional Islamic State of Afghanistan and UNHCR (2002)	54
	3.15	The Islamic Transitional State of Afghanistan, the Government of the Netherlands	01
	0110	and UNHCR (2003)	54
	3.16	Republic of Zambia, Republic of Rwanda and UNHCR (2003)	55
	3.17	Islamic Republic of Iran, the Islamic Transitional State of Afghanistan	20
		and UNHCR (2003)	55
	3.18	Republic of Rwanda, Republic of Uganda and UNHCR (2003)	55
	3.19	Republic of Burundi, Republic of Rwanda and UNHCR (2005)	56
	3.20	Republic of Sudan, Democratic Republic of Congo and UNHCR (2006)	56

4	Unite	ed Nations High Commissioner for Refugees (UNHCR)	56
	4.1	EXCOM Conclusion No. 18 – Voluntary Repatriation (1980)	56
	4.2	EXCOM Conclusion No. 40 – Voluntary Repatriation (1985)	57
	4.3	EXCOM Conclusion No. 101 – Legal Safety Issues in the Context of Voluntary	
		Repatriation of Refugees (2004)	58
	4.4	UNHCR Inter-Office Memorandum No. 104/2001, UNHCR Field Office	
		Memorandum No. 101/2001 – Voluntary Repatriation and the Right to Adequate	
		Housing	61
	4.5	Global Consultations on International Protection (2002)	71
5	UN S	ecurity Council	73
	5.1	Resolution 237 – The Situation in the Middle East (1967)	73
	5.2	Resolution 242 – Middle East (1967)	74
	5.3	Resolution 361 – Cyprus (1974)	74
	5.4	Resolution 674 – Kuwait (1990)	74
	5.5	Resolution 687 – Kuwait (1991)	74
	5.6	Resolution 752 – Yugoslavia (1992)	75
	5.7	Resolution 787 – Bosnia and Herzegovina (1992)	76
	5.8	Resolution 820 – Yugoslavia (1993)	76
	5.9	Resolution 853 – Azerbaijan (1993)	76
	5.10	Resolution 876 – Georgia (1993)	76
	5.11	Resolution 947 – Croatia (1994)	76
	5.12	Resolution 971 – Georgia/Abkhazia (1995)	76
	5.13	Resolution 999 – Tajikistan (1995)	77
	5.14	Resolution 1009 – Croatia (1995)	77
	5.15	Resolution 1019 – Croatia (1995)	77
		Resolution 1036 – Georgia/Abkhazia (1996)	77
		Resolution 1079 – Croatia (1996)	78
	5.18	Resolution 1088 – Bosnia and Herzegovina (1996)	78
		Resolution 1120 – Croatia (1997)	78
		Resolution 1145 – Croatia (1997)	79
		Resolution 1199 – Kosovo (1998)	79
		Resolution 1244 – Kosovo (1999)	80
	5.23	Resolution 1287 – Abkhazia and the Republic of Georgia	
		(2000)	80
		Resolution 1339 – Georgia (2001)	81
		Resolution 1357 – Bosnia and Herzegovina (2001)	81
		Resolution 1364 – Georgia (2001)	81
	5.27	Resolution 1393 – Georgia (2002)	82
	5.28	Resolution 1427 – Georgia (2002)	82
	5.29	Resolution 1462 – Georgia (2003)	82
	5.30	Resolution 1491 – Bosnia and Herzegovina (2003)	83
	5.31	Resolution 1494 – Georgia (2003)	83
	5.32	Resolution 1524 – Georgia (2004)	84
	5.33	Resolution 1545 – Burundi (2004)	84
	5.34 5.25	Resolution 1551 – Bosnia and Herzegovina (2004)	85
	5.35	Resolution 1556 – Sudan (2004) Resolution 1575 – Rospie and Herzogovine (2004)	85
	5.36	Resolution 1575 – Bosnia and Herzegovina (2004)	85
	5.37 5.38	Resolution 1582 – Georgia (2005) Resolution 1615 – Coorgia (2005)	85
	5.38	Resolution 1615 – Georgia (2005)	86

6	UN C	General Assembly	87
	6.1	Resolution 8 – Question of Refugees (1946)	87
	6.2	Resolution 194 – United Nations Conciliation Commission for Palestine, Protection	
		and a Durable Solution for Palestinian Refugees (1948)	87
	6.3	Resolution 394 – Palestine: Progress Report of the United Nations Conciliation	
		Commission for Palestine; Repatriation or Resettlement of Palestine Refugees and	
		Payment of Compensation Due to Them (1950)	89
	6.4	Resolution 428 – Statute of the Office of the United Nations High Commissioner for	
		Refugees (1950)	89
	6.5	Addendum to Definition of a "Refugee" under Paragraph 11 of the General Assembly	
		Resolution (11 December 1948) (1951)	89
	6.6	Resolution 1388 – Report of the United Nations High Commissioner	
		for Refugees (1959)	91
	6.7	Resolution 1390 – World Refugee Year (1959)	91
	6.8	Resolution 1672 – Refugees from Algeria in Morocco and Tunisia (1961)	91
	6.9	Resolution 3212 – Question of Cyprus (1974)	92
	6.10	Resolution 3236 - Question of Palestine (1974)	92
	6.11	Resolution 35/124 – International Cooperation to Avert New Flows	
		of Refugees (1980)	92
	6.12	Resolution 36/146 – United Nations Relief and Works Agency for Palestine Refugees	
		in the Near East (1981)	92
	6.13	Resolution 36/148 – International Co-Operation to Avert New Flows	
		of Refugees (1981)	93
	6.14	Resolution 37/120 – United Nations Relief and Works Agency for Palestine Refugees	
		in the Near East (1982)	95
	6.15	Resolution 40/165 – United Nations Relief and Works Agency for Palestine Refugees	
		in the Near East (1985)	95
	6.16	Resolution 48/117 – International Conference on Central American Refugees (1993)	96
	6.17	Resolution 48/118 – Assistance to Refugees, Returnees and Displaced Persons in	
		Africa (1993)	96
	6.18	Resolution 48/152 – Situation of Human Rights in Afghanistan (1993)	97
	6.19	Resolution 49/23 – Emergency International Assistance for a Solution to the	
		Problem of Refugees, the Restoration of Total Peace, Reconstruction and	
		Socio-Economic Development in War-Stricken Rwanda (1994)	98
	6.20	Resolution 49/43 – The Situation in the Occupied Territories of Croatia (1994)	98
	6.21	Resolution 49/196 – Situation of Human Rights in the Republic of Bosnia and	
		Herzegovina, the Republic of Croatia and the Federal Republic of Yugoslavia	
		(Serbia and Montenegro) (1995)	98
	6.22	Resolution 50/182 – Human Rights and Mass Exoduses (1995)	99
	6.23	Resolution 51/114 – Situation of Human Rights in Rwanda (1996)	99
	6.24	Resolution 51/126 – Persons Displaced as a Result of the June 1967 and Subsequent	
		Hostilities (1996)	99
	6.25	Resolution 55/153 – Nationality of Natural Persons in Relation to the Succession of	
		States (2000)	100
	6.26	Resolution 59/117 – Assistance to Palestine Refugees (2004)	100
	6.27	Resolution 59/170 – Office of the United Nations High Commissioner	
		for Refugees (2004)	100
	6.28	Resolution 59/172 – Assistance to Refugees, Returnees and Displaced Persons in	
		Africa (2004)	101
	6.29	Resolution – 60/100 – Assistance to Palestinian Refugees (2005)	101

	6.30	Resolution 60/101 – Persons Displaced as a Result of the June 1967 and Subsequent	
		Hostilities (2005)	102
	6.31	Resolution 60/103 – Palestine Refugees' Properties and Their Revenues (2005)	102
	6.32	Resolution 60/147 – Basic Principles and Guidelines on the Right to a Remedy and	
		Reparation for Victims of Gross Violations of International Human Rights Law and	
		Serious Violations of International Humanitarian Law (2005)	103
	6.33	Resolution 60/183 – Permanent Sovereignty of the Palestinian People in the	
		Occupied Palestinian Territory, Including East Jerusalem, and of the Arab	
		Population in the Occupied Syrian Golan over Their Natural Resources (2005)	104
7	UNC	0	105
	7.1	Resolution 1993/77 – Forced Evictions (1993)	105
	7.2	Resolution 1997/29 – The Right to Restitution, Compensation and Rehabilitation for	
		Victims of Grave Violations of Human Rights and Fundamental Freedoms (1997)	107
	7.3	The Guiding Principles on Internal Displacement (1998)	107
	7.4	Resolution 1999/33 – The Right to Restitution, Compensation and Rehabilitation for	
		Victims of Grave Violations of Human Rights and Fundamental Freedoms (1999)	108
	7.5	Report of the Independent Expert on the Right to Restitution, Compensation and	
		Rehabilitation for Victims of Grave Violations of Human Rights and Fundamental	
		Freedoms, Mr. M. Cherif Bassiouni, Submitted Pursuant to Commission on Human	
		Rights Resolution 1998/43 (1999) (Excerpts)	109
	7.6	Resolution 2000/62 – Basic Principles and Guidelines on the Right to a Remedy and	
		Reparation for Victims of Violations of International Human Rights and	
		Humanitarian Law (2000)	112
	7.7	Resolution 2003/34 – The Right to Restitution, Compensation and Rehabilitation for	
		Victims of Grave Violations of Human Rights and Fundamental Freedoms (2003)	113
	7.8	Resolution 2004/28 – Prohibition of Forced Evictions (2004)	114
	7.9	Resolution 2004/34 – The Right to Restitution, Compensation and Rehabilitation for	
		Victims of Grave Violations of Human Rights and Fundamental Freedoms (2004)	115
	7.10	Resolution 2004/55 – Internally Displaced Persons (2004)	116
	7.11	Resolution 2005/46 – Internally Displaced Persons (2005)	117
8	UN S	ub-Commission on the Protection and Promotion of Human Rights	117
	8.1	Study Concerning the Right to Restitution, Compensation and Rehabilitation for	
		Victims of Gross Violations of Human Rights and Fundamental Freedoms – Final	
		Report Submitted by Mr. Theo van Boven, Special Rapporteur (1993) (Excerpts)	117
	8.2	Resolution 1994/24 – The Right to Freedom of Movement (1994)	119
	8.3	Resolution 1995/13 – The Right to Freedom of Movement (1995)	120
	8.4	Resolution 1997/31 – The Right to Return (1997)	120
	8.5	Resolution 1998/9 – Forced Evictions (1998)	120
	8.6	Resolution 1998/26 – Housing and Property Restitution in the Context of the Return	
		of Refugees and Internally Displaced Persons (1998)	122
	8.7	The Return of Refugees' or Displaced Persons' Property – Working Paper Submitted	
	011	by Mr. Paulo Sérgio Pinheiro (2002)	123
	8.8	Resolution 2002/7 – Housing and Property Restitution in the Context of Refugees	120
	010	and Other Displaced Persons (2002)	132
	8.9	Resolution 2002/30 – The Right to Return of Refugees and Internally Displaced	102
	0.0	Persons (2002)	133
	8.10	Housing and Property Restitution in the Context of the Return of Refugees and	100
	0.10	Internally Displaced Persons – Preliminary Report of the Special Rapporteur, Paulo	
		Sérgio Pinheiro (2003)	135
		orgio i mitero (2003)	100

	8.11	Housing and Property Restitution in the Context of the Return of Refugees and Internally Displaced Persons – Progress Report of the Special Rapporteur, Paulo	
		Sérgio Pinheiro (2004)	147
	8.12	Resolution 2004/2 – Housing and Property Restitution (2004)	157
	8.13	Housing and Property Restitution in the Context of the Return of Refugees and	
		Internally Displaced Persons – Final Report of the Special Rapporteur, Paulo Sérgio	
		Pinheiro (2005)	158
	8.14	Resolution 2005/21 – Housing and Property Restitution for Refugees and Displaced	
		Persons (2005)	160
9	UN H	luman Rights Treaty Bodies	. 160
	9.1	Committee on Economic, Social and Cultural Rights	160
		9.1.1 General Comment 4 – The Right to Adequate Housing (1991)	160
		9.1.2 General Comment 7 – Forced Evictions and the Right to Adequate Housing	
		(1997)	164
	9.2	Committee on the Elimination of Racial Discrimination	168
		9.2.1 General Recommendation 22 – Article 5 and Refugees and Displaced Persons	
		(1996)	168
		9.2.2 General Recommendation 23 – Rights of Indigenous Peoples (1997)	168
	9.3	Human Rights Committee	169
		9.3.1 General Comment 16 – The Right to Respect of Privacy, Family, Home and	
		Correspondence, and Protection of Honour and Reputation (Art. 17)(1988)	169
		9.3.2 General Comment 27 – Freedom of Movement (1999)	171
10	Addi	ional Standards	. 174
	10.1	The Cairo Declaration of Principles of International Law on Compensation to	
		Refugees (1993)	174
	10.2	The Practice of Forced Evictions: Comprehensive Human Rights Guidelines on	
		Development-Based Displacement (1997)	176
PAR	T TWO	. HOUSING AND PROPERTY RESTITUTION STANDARDS – REGIONAL	181
1	Afric	a	. 181
	1.1	OAU Convention Governing the Specific Aspects of Refugee Problems	
		in Africa (1974)	181
2	Ame	icas	. 181
	2.1	American Convention on Human Rights (1969)	181
	2.2	Cartagena Declaration on Refugees (1984)	182
3	Euro	ре	. 182
	3.1	European Convention for the Protection of Human Rights and Fundamental	
		Freedoms (1950)	182
4	Midd	le East	. 183
	4.1	The Cairo Declaration on Human Rights in Islam (1990)	183
	4.2	Declaration on the Protection of Refugees and Displaced Persons in the Arab	100
		World (1992)	184
	4.3	Arab Charter on Human Rights (1994)	186
DAD	ттир	EE. HOUSING AND PROPERTY RESTITUTION LAWS AND STANDARDS – NATIONAL	187
1		inistan	
1	1.1	Decree on Dignified Return of Refugees (2001)	187
	1.1 1.2	Decree 89 of the Head of the Transitional Islamic State of Afghanistan, Regarding the	107
	1.2	Creation of a Special Property Disputes Resolution Court (2003)	188
		Greation of a special roperty Disputes hesolution Court (2003)	100

2	Alban	ia	. 189
	2.1	Law No. 9235 on Restitution and Compensation of Property (2004)	189
3	Arme	nia	
	3.1	On the Legal and Socio-economic Guarantees for the Persons Who Had Been	
		Forcibly Displaced from the Republic of Azerbaijan in 1988–1992 and Have Acquired	
		the Citizenship of the Republic of Armenia (2000)	197
4	Azerb	aijan	. 197
	4.1	Law on the Social Protection of Internally Displaced Persons and Persons with Equal	105
_		Status (1999)	197
5		a-Herzegovina	198
	5.1	Law on the Cessation of Application of the Law on the Use of Abandoned Property	
	- 0	(Republika Srpska)(1996)	198
	5.2	PIC Sintra Declaration (1997)	204
	5.3	Office of the High Commissioner, Federation Forum, Chairman's Conclusions (1997)	205
	5.4	Law on Implementation of the Decisions of the Commission for Real Property Claims of Displaced Persons and Refugees (Republika Srpska) (1999)	206
	5.5	Law on Implementation of the Decisions of the Commission for Real Property	200
	5.5	Claims of Displaced Persons and Refugees (Federation BiH) (1999)	209
	5.6	Property Law Implementation Plan (PLIP) Inter-Agency Framework Document	203
	5.0	(2000)	212
	5.7	A New Strategic Direction: Proposed Ways Ahead for Property Law Implementation	
		in a Time of Decreasing IC [International Community] Resources (2002)	222
	5.8	PLIP Municipal Guidelines for Substantial Completion of Property Law	
		Implementation (2003)	229
	5.9	Law on the Cessation of the Application of the Law on Temporary Abandoned Real	
		Property Owned by Citizens (Federation BiH) (2003)	230
	5.10	Law on the Cessation of the Application of the Law on Abandoned Apartments	
		(Federation BiH) (2003)	237
	5.11	Law on the Cessation of Application of the Law on the Use of Abandoned Property	
		(Republika Srpska) (2003)	247
	5.12	Law on Implementation of the Decisions of the Commission for Real Property	
		Claims of Displaced Persons and Refugees (Federation of BiH) (2003)	263
	5.13	Law on Implementation of the Decisions of the Commission for Real Property	
	5 1 4	Claims of Displaced Persons and Refugees (Republika Srpska) (2003)	266
	5.14	Commission for Real Property Claims of Displaced Persons and Refugees – Book of Regulations on Confirmation of Occupancy Rights of Displaced Persons and	
		Refugees (2003)	269
	5.15	Commission for Real Property Claims of Displaced Persons and Refugees – Book of	203
	5.15	Regulations on the Conditions and Decision Making Procedure for Claims for	
		Return of Real Property of Displaced Persons and Refugees (2003)	277
6	Bulga	ria	
0	6.1	Restitution of Nationalised Real Property Act (1992)	288
	6.2	Decree No. 60 on Applying Article 1, Paragraph 2 of the Restitution of Nationalised	200
	0.2	Real Property Act (1992)	290
7	Color	nbia	
'			
0	7.1	Decree No. 2007 (2001)	291
8			
	8.1	Republic of Estonia Principles of Ownership Reform Act (1991)	294

9	Geor	gia	. 306
	9.1 9.2	Law of the Republic of Georgia Concerning Internally Displaced People (1996) Law of Georgia on Property Restitution and Compensation on the Territory of	306
		Georgia for the Victims of Conflict in Former South Ossetia District (2006, Draft)	306
10	Germ 10.1	any	. 317
		Issues (1990)	317
11	-		318
	11.1	Coalition Provisional Authority Order Number 6 – Eviction of Persons Illegally	
	11.0	Occupying Public Buildings (2003)	318
	11.2	Coalition Provisional Authority Regulation Number 4 – Establishment of the Iraqi Property Reconciliation Facility (2004)	210
	11.3	Coalition Provisional Authority Regulation Number 12 – Iraqi Property Claims	319
	11.5	Commission (2004)	320
	11.4	Statute of the Commission for the Resolution of Real Property Disputes (2006)	334
12	Koso	νο	340
	12.1	UNMIK Regulation 1999/10 on the Repeal of Discriminatory Legislation Affecting	
		Housing and Rights in Property (1999)	340
	12.2	UNMIK Regulation 1999/23 on the Establishment of the Housing and Property	
		Directorate and the Housing and Property Claims Commission (1999)	341
	12.3	UNMIK Regulation 2000/60 on Residential Property Claims and the Rules of	
		Procedure and Evidence of the Housing and Property Directorate and the Housing and Property Claims Commission (2000)	342
13	Rom	ania	. 353
15		Law No. 112 Regulating the Legal Status of Certain Residential Property (1995)	353
14		nda	. 354
14	14.1	Ministerial Order No. 01 – Regarding the Temporary Management of Land Property	. 554
	14.1	(1996)	354
15	South	n Africa	. 355
	15.1	Restitution of Land Rights Act 22 (1994)	355
		South Africa Constitution (1996)	356
16	Tajiki	stan	. 357
	16.1	The Law of the Republic of Tajikistan on Forced Migrants (1994)	357
	16.2	Resolution No. 542 – Additional Measures Facilitating the Return of	
		Refugees-Citizens of the Republic of Tajikistan and Forced Migrants to the Places of	
		Permanent Residence and Their Social and Legal Protection (1995)	358
PAR	T FOU	R. HOUSING AND PROPERTY RESTITUTION CASE LAW	359
1	Perm	anent International Court of Justice	. 359
	1.1	The Factory at Chórzow (Indemnity) Case (1928)	359
2	Inter	national Court of Justice	. 361
	2.1	Advisory Opinion on the Legal Consequences of the Construction of a Wall in the	
		Occupied Palestinian Territory (2004)	361
3	Hum	an Rights Committee	. 362
	3.1	Simunek, Hastings, Tuzilova and Prochazka v. The Czech Republic (1995)	362
	3.2	Dr. Karel Des Fours Walderode v. The Czech Republic (1996)	369

	3.3	Josef Frank Adam v. The Czech Republic (1996)	373
	3.4	Ms. Eliska Fábryová v. The Czech Republic (1997)	378
	3.5	Robert Brok v. The Czech Republic (2001)	380
	3.6	Miroslav Blazek, George A. Hartman and George Krizek v. The Czech Republic (2001)	384
4	Euro	pean Court of Human Rights	388
	4.1	Sporrong and Lönnroth v. Sweden (1982)	388
	4.2	Lithgow and Others v. United Kingdom (1986)	394
	4.3	Papamichalopoulos and Others v. Greece (1993)	407
	4.4	Loizidou v. Turkey (1996) and Damages (1998)	412
	4.5	Zubani v. Italy (1996)	423
	4.6	Akdivar and Others v. Turkey (1996) and Damages (1998)	427
	4.7	Guillemin v. France (1997)	434
	4.8	Brumărescu v. Romania (1999)	437
	4.9	Bilgin v. Turkey (2000)	441
	4.10	Zwierzyński v. Poland (2001)	443
	4.11	Cyprus v. Turkey (2001)	448
	4.12	Dulaş v. Turkey (2001)	458
	4.13	Orhan v. Turkey (2002)	461
	4.14	Zvolský and Zvolská v. The Czech Republic (2003)	461
	4.15	Jasiūnienė v. Lithuania (2003)	466
	4.16	Pincová and Pinc v. The Czech Republic (2003)	469
	4.17	Wittek v. Germany (2003)	477
	4.18	Jantner v. Slovakia (2003)	483
	4.19	Valová, Slezák and Slezák v. Slovakia (2004)	486
	4.20	Broniowski v. Poland (2004)	489
	4.21	Doğan and Others v. Turkey (2004)	508
	4.22	Străin and Others v. Romania (2005)	518
	4.23	Xenides-Arestis v. Turkey (2006)	524
	4.24	Blečić v. Croatia (2006)	529
PAF	T FIVE	. USEFUL RESOURCES ON HOUSING AND PROPERTY RESTITUTION	537
1		ral Resources	537
2		try/Regional Resources	
3	Useft	ıl Web Sites	545
Ind	ex		547

Acknowledgments

Although the preparation of a book with such sparse narrative as this may seem a quick and easy task, the ultimate completion of this volume – as they somehow always do – took far longer than originally envisaged. As it turned out, accessing, reviewing, selecting, and finally bringing together some of the literally thousands of possible standards on restitution issues was far more complex than initially foreseen at the time of the enthusiastic outset of this process.

Putting together a volume of this size and scope alone would have been truly daunting, and without the generous contribution of considerable time and energy by many friends and colleagues, it surely would have never seen the light of day.

First off, special thanks are due to Jessica Marasovic of the Centre on Housing Rights and Evictions (COHRE), who helped me itemise and access many of the standards found in this volume. Her computer-savvy ways and constantly upbeat spirit greatly assisted throughout the preparation process. Thanks also to Dima Yared for her efforts and to Rhodri Williams, who provided some of the more difficult-to-access texts concerning Bosnia-Herzegovina.

I am very thankful to the Board of Directors of COHRE, most notably our chairperson, John

Packer, for approving my request to take a sabbatical leave from my position as executive director of COHRE, without which I may have never found the time to complete this and other works.

And thank you, too, to the many extraordinary people whom I met and who assisted me in so many important ways along the arduous and far too lengthy road that was required to bring this book to fruition. I was fortunate enough to have worked on this volume in a number of my favourite countries – Australia, Cambodia, Canada, Palestine, Switzerland, and Thailand – and offer anonymous thanks to the dozens of friends and other colleagues in these and other places who helped out in important but often unknown ways.

I am especially grateful to John Berger at Cambridge University Press for his support for this project and his extraordinary patience in waiting so calmly for it to come to an end.

And, as always, my deepest thanks go to Kirsten Young – The Harling – inspiration, lifesaver, soulmate, partner, source of my joy, light of my day.

> Scott Leckie Bangkok, 6 August 2006

Preface

This volume contains some 240 of the most important international, regional, and national standards and judicial decisions recognising and addressing the many complexities associated with the housing, land, and property restitution rights of refugees and displaced persons. Many additional laws, cases, and materials on these themes are also in place throughout the world, but the selection found here attempts to provide an accurate and reasonably detailed picture of the current state of affairs with respect to these rights set within their various normative contexts.

The texts included within this book - that can be seen as collectively forming the normative basis for the right of refugees and displaced persons to have restored to them the housing, land, and property rights they held at the time of their displacement - stretch back ten decades to the Hague Conventions of 1907. Throughout the past century, numerous standards have been proposed, discussed, and ultimately approved on restitution, both internationally and at the national level. Over time, these documents have become increasingly specific, in the process refining the precise rights to housing, land, and property restitution enjoyed by those forced by circumstances beyond their control to flee their homes and lands.

In many respects, 2005 was witness to a high point of international standard-setting on these issues, with two vitally important new restitution standards being approved by United Nations (UN) bodies that considerably augment the housing, land, and property restitution rights of refugees and displaced persons. In August 2005, the 'Pinheiro' Principles on Housing and Property Restitution for Refugees and Displaced Persons (see Section 1.15 for the full text) were approved by the UN Sub-Commission on the Protection and Promotion of Human Rights. The 'Pinheiro' Principles provide the most comprehensive and consolidated international norm outlining the rights of refugees and displaced persons to have restored to them all of the housing, land, and property rights they held before their displacement.

In December 2005, the UN General Assembly adopted the *Basic Principles and Guidelines on the Right to a Remedy and Reparation for Victims of Violations of International Human Rights and Humanitarian Law* (see Section 1.16 for the full text), which further strengthens the rights of those forced from their homes to achieve a proper remedy for these abuses through the process of restitution. Together, these two new norms exemplify both the level of specificity that has been reached in the international recognition of these rights, as well as indicating the considerable seriousness now accorded these rights by the global community.

While much of the twentieth century was rather unkind to the prospect of housing, land, and property restitution, during the past two decades millions of people throughout the world have been able to formally exercise their housing, land, and property restitution rights and return home, from Tajikistan to Kosovo, from Mozambique to Liberia, and from Bosnia-Herzegovina to South Africa and beyond. Many of the standards contained in this volume were key ingredients in the creation of conditions that led to their eventual return home. These standards have not, alas, been taken seriously in every context they should have been applied, and as a result many tens of millions of refugees and displaced persons with existing, legally recognised, and fully legitimate restitution rights are unable to exercise them because of resistance by States and others responsible for the original displacement, many of which, in fact, are due to arbitrary, discriminatory, and unfair national laws specifically designed to prevent the exercise of housing, land, and property restitution rights.

A Note on the Contents

This volume is intended to be a reference tool and consolidated source of *selected* international, regional, and national legal standards, nonlegal materials, and case law concerning the housing, land, and property restitution rights of refugees and displaced persons. This book is designed to assist those working with people asserting restitution rights – lawyers, refugee and IDP advocates, government officials, UN staff members, fieldworkers, and others – to easily access various types of standards recognising these rights within a single guide.

The volume does not provide analysis of the various standards, nor does it explain their nature, legal standing, or relative normative value. For those readers wishing to explore these issues, they may wish to review Part Five, which provides the citations of a wide range of books, articles, and other analyses of housing and property restitution issues. Most, if not all, of these standards are accessible on the Internet, and readers wishing to reference or otherwise use any of these norms are urged to examine these online to ensure that any updates, amendments, repeals, or other changes are adequately understood.

With one or two notable exceptions, every effort has been made to not include texts that have subsequently been repealed or substantially amended. However, given the frequent changes incorporated into domestic legislation, readers are again urged to confirm the current status of the national laws included in the volume before referencing them in any official manner. In one instance - that of Iraq even though the standard contained in Section 12.4 resulted in the repeal of one of the regulations preceding it, it was felt important to retain the repealed standard to give readers a sense of the type of restitution regulations occupying powers, in this instance the United States, have proclaimed during their contentious occupation of this sovereign nation.

In the interests of preventing repetition and excessive length, considerable effort was made during the compilation of this volume to limit the number of particular types of standards, most notably relevant General Assembly resolutions and nonrefugee-specific national restitution legislation. The General Assembly, for instance, very rightly adopts essentially the same series of resolutions each year in support of the housing, land, and property restitution rights of Palestinian refugees, but this volume only includes several of the more important pronouncements in this regard, including the most recent such series of resolutions approved in 2005.

In terms of national law, a concerted effort was made to only include a select few examples of domestic legislation regulating restitution efforts in formerly communist countries. Restitution programmes were carried out in many such countries; however, given that the emphasis in this book is on the housing, land, and property restitution rights held explicitly by refugees and displaced persons, it was felt that noting *some* of the restitution laws in Eastern Europe and elsewhere would suffice in revealing the nature of these laws, how they are formulated, and, ultimately, how they differ with and or are similar to restitution laws specifically designed to benefit refugees and the displaced.

A Note on Organisation

Part One includes ten sections, each of which contain *international* texts addressing a wide variety of restitution themes. Section 1 includes sixteen international standards under international humanitarian law, international human rights law, international criminal law, international refugee law, the international law of state responsibility, and specific restitution standards.

Section 2 contains excerpts from fourteen peace agreements concluded since 1991, which explicitly or implicitly recognise housing, land, and property restitution rights for those displaced as a result of the conflict concerned. Peace agreements do not yet systematically address these issues, yet a pattern is emerging whereby constructively addressing housing, land, and property rights concerns within a rights-based peace-building process is increasingly seen as a major component of sustainable peace, reconciliation, economic viability, and overall political stability.

Twenty voluntary repatriation agreements concluded between United Nations High Commissioner for Refugees (UNHCR) and various governments are included in Section 3. Such agreements increasingly include reference to housing, land, and property issues as the international community comes to embrace the notion that successful

Preface

peace-building can only take place when these issues are properly and justly addressed.

UNHCR's role in strengthening restitution norms for refugees has been a vital one and, in recognition of this, beyond the agreements found in Section 3, five additional UNHCR texts are included in Section 4.

Section 5 includes full texts and excerpts of thirty-nine resolutions adopted by the UN Security Council since 1967 that affirm and reaffirm the crucial nature of protecting the housing, land, and property restitution rights of refugees and others forcibly displaced from their homes and lands.

Similarly, the UN General Assembly has addressed restitution issues since its inception, and as a result, scores of resolutions have been approved by this body, thirty-three of which are included in Section 6 of this volume.

The UN human rights machinery has also become increasingly involved in efforts to promote the housing, land, and property rights of refugees and displaced persons, and various resolutions and other standards approved by the Commission on Human Rights (now "Human Rights Council"), Sub-Commission on the Protection and Promotion of Human Rights, and several human rights treaty bodies are included in Sections 7–9. Section 10 contains two relevant standards approved by expert bodies, which address or are relevant to restitution themes.

Part Two provides several *regional* standards that address housing, land, and property restitution issues in Africa, the Americas, Europe, and the Middle East.

Part Three of the volume contains forty *national* laws and other statements enshrining housing, land, and property restitution rights. Laws from the following sixteen countries are provided: Afghanistan, Albania, Armenia, Azerbaijan, Bosnia-Herzegovina, Bulgaria, Colombia, Estonia, Georgia, Germany, Iraq, Kosovo, Romania, Rwanda, South Africa, and Tajikistan.

Part Four contains selected case law from four adjudicating bodies: the Permanent Court of International Justice, the International Court of Justice, the UN Human Rights Committee, and the European Court on Human Rights. Readers should note that although most of the cases included essentially support the contention that refugees, displaced persons, or other human rights victims possess housing, land, and property rights, in some instances cases have been included that are more nuanced or even nonsupportive of such perspectives.

Finally, Part Five provides a selected bibliography of various publications addressing different dimensions of the housing, land, and property restitution question.

Readers should note that the various documents found in this book have been organised in descending order, with each section beginning with the oldest documents and ending with the newest.

A Note on What Is Not Included

Although not as numerous as laws in support of restitution rights, many countries have adopted legislation consciously designed to *prevent* the exercise of housing, land, and property rights by refugees and displaced persons. Israel's absentee property laws, Bhutan's nationality laws, housing laws in Croatia, Rwandan legislation on refugee return, and many other pieces of existing law remain in place at the national level in spite of the clear preponderance of normative evidence found in this volume that those displaced from their homes have a right to return to and repossess those homes.

The texts of such laws are not included in the present volume, although the idea of systematically compiling these antirestitution laws into a separate volume may have considerable merit, and anyone wishing to make a positive contribution to the restitution question may well consider undertaking such an exercise.

A Note on Sources

As noted, most of the documents included within this volume are available electronically on the Internet, with the exception of voluntary repatriation agreements and some of the national laws outlined here. Full texts of those available online can best be accessed through the following Web sites:

 http://un.org for Part One – Section 1 (International Standards), Section 5 (UN Security Council), Section 6 (UN General Assembly), Section 7 (UN Commission on Human Rights), Section 8 (UN Sub-Commission on the Protection and Promotion of Human Rights), and Section 9 (UN Human Rights Treaty Bodies) and Part Four – Sections 2 and 3;

- (2) http://www.usip.org/library/pa.html for Part One – Section 2 (Peace Agreements);
- (3) http://www.unhcr.org/cgi-bin/texis/vtx/rsd for Part One – Section 3 (Voluntary Repatriation

Agreements) and Section 4 (UN High Commissioner for Refugees);

- (4) http://www.worldcourts.com for Part Four Section 1 (Permanent Court of International Justice); and
- (5) http://www.echr.coe.int/ECHR/EN/Header/ Case-Law/HUDOC/HUDOC+database for Part Four – Section 4 (European Court of Human Rights).

PART ONE

Housing and Property Restitution Standards – International

1. INTERNATIONAL STANDARDS

INTERNATIONAL HUMANITARIAN LAW

1.1. Hague Convention (IV) Respecting the Laws and Customs of War on Land $(1907)^1$

[...]

Section II – Hostilities – Chapter I – Means of Injuring the Enemy

[...]

Article 23

In addition to the prohibitions provided by special Conventions, it is especially forbidden

[...]

(g) To destroy or seize the enemy's property, unless such destruction or seizure be imperatively demanded by the necessities of war;

(h) To declare abolished, suspended, or inadmissible in a court of law the rights and actions of the nationals of the hostile party...

[...]

Article 25

The attack or bombardment, by whatever means, of towns, villages, dwellings, or buildings which are undefended is prohibited.

[...]

Article 28

The pillage of a town or place, even when taken by assault, is prohibited.

[...]

1.2. Geneva Convention (IV) Relative to the Protection of Civilian Persons in Time of War $(1949)^2$

[...]

Article 33

No protected person may be punished for an offence he or she has not personally committed. Collective penalties and likewise all measures of intimidation or of terrorism are prohibited.

Pillage is prohibited.

Reprisals against protected persons and their property are prohibited.

[...]

Article 45

Protected persons shall not be transferred to a Power which is not a party to the Convention.

This provision shall in no way constitute an obstacle to the repatriation of protected persons, or to their return to their country of residence after the cessation of hostilities.

Article 46

In so far as they have not been previously withdrawn, restrictive measures taken regarding

¹ Adopted on 18 October 1907 and entered into force on 26 January 1910.

² Adopted on 12 August 1949 by the Diplomatic Conference for the Establishment of International Conventions for the Protection of Victims of War, held in Geneva from 21 April to 12 August 1949 and entered into force on 21 October 1950.

protected persons shall be cancelled as soon as possible after the close of hostilities.

Restrictive measures affecting their property shall be cancelled, in accordance with the law of the Detaining Power, as soon as possible after the close of hostilities.

[...]

Article 49

Individual or mass forcible transfers, as well as deportations of protected persons from occupied territory to the territory of the Occupying Power or that of any other country, occupied or not, are prohibited, regardless of their motive.

Nevertheless, the occupying Power may undertake total or partial evacuation of a given area if the security of the population or imperative military reasons so demand. Such evacuations may not involve the displacement of protected persons outside the bounds of the occupied territory except when for material reasons it is impossible to avoid such displacement. Persons thus evacuated shall be transferred back to their homes as soon as hostilities in the area in question have ceased.

The Occupying Power undertaking such transfers or evacuations shall ensure, to the greatest practicable extent, that proper accommodation is provided to receive the protected persons, that the removals are effected in satisfactory conditions of hygiene, health, safety and nutrition, and that members of the same family are not separated.

The Protecting Power shall be informed of any transfers and evacuations as soon as they have taken place.

The Occupying Power shall not detain protected persons in an area particularly exposed to the dangers of war unless the security of the population or imperative military reasons so demand.

The Occupying Power shall not deport or transfer parts of its own civilian population into the territory it occupies.

[...]

Article 53

Any destruction by the Occupying Power of real or personal property belonging individually or collectively to private persons, or to the State, or to other public authorities, or to social or cooperative organizations, is prohibited, except where such destruction is rendered absolutely necessary by military operations.

[...]

Article 147

Grave breaches...shall be those involving any of the following acts, if committed against persons or property protected by the present Convention:...extensive destruction and appropriation of property, not justified by military necessity and carried out unlawfully and wantonly.

[...]

1.3. Protocol Additional to the Geneva Conventions of 12 August 1949, and Relating to the Protection of Victims of International Armed Conflicts (Protocol 1) (1977)³

[...]

Chapter III - Civilian Objectives

Article 52 - General Protection of Civilian Objects

1. Civilian objects shall not be the object of attack or of reprisals. Civilian objects are all objects which are not military objectives as defined in paragraph 2.

2. Attacks shall be limited strictly to military objectives. In so far as objects are concerned, military objectives are limited to those objects which by their nature, location, purpose or use make an effective contribution to military action and whose total or partial destruction, capture or neutralization, in the circumstances ruling at the time, offers a definite military advantage.

3. In case of doubt whether an object which is normally dedicated to civilian purposes, such as a place of worship, a house or other dwelling or a school, is being used to make an effective contribution to military action, it shall be presumed not to be so used.

[...]

³ Adopted on 8 June 1977 and entered into force 7 December 1978 in accordance with Article 95.

Section III – Treatment of Persons in the Power of a Party to the Conflict

Chapter I – Field of Application and Protection of Persons and Objects

Article 73 - Refugees and Stateless Persons

Persons who, before the beginning of hostilities, were considered as stateless persons or refugees under the relevant international instruments accepted by the Parties concerned or under the national legislation of the State of refuge or State of residence shall be protected persons within the meaning of Parts I and III of the Fourth Convention, in all circumstances and without any adverse distinction.

[...]

1.4. Protocol Additional to the Geneva Conventions of 12 August 1949, and Relating to the Protection of Victims of Non-International Armed Conflicts (Protocol II) (1977)⁴

[...]

Article 13 - Protection of the Civilian Population

1. The civilian population and individual civilians shall enjoy general protection against the dangers arising from military operations. To give effect to this protection, the following rules shall be observed in all circumstances.

2. The civilian population as such, as well as individual civilians, shall not be the object of attack. Acts or threats of violence the primary purpose of which is to spread terror among the civilian population are prohibited.

3. Civilians shall enjoy the protection afforded by this Part, unless and for such time as they take a direct part in hostilities.

[...]

Article 17 – Prohibition of Forced Movement of Civilians

1. The displacement of the civilian population shall not be ordered for reasons related to the conflict

unless the security of the civilians involved or imperative military reasons so demand. Should such displacements have to be carried out, all possible measures shall be taken in order that the civilian population may be received under satisfactory conditions of shelter, hygiene, health, safety and nutrition.

2. Civilians shall not be compelled to leave their own territory for reasons connected with the conflict.

[...]

INTERNATIONAL HUMAN RIGHTS LAW

1.5. Universal Declaration of Human Rights (1948)⁵

[...]

Article 8

Everyone has the right to an effective remedy by the competent national tribunals for acts violating the fundamental rights granted him by the constitution or by law.

[...]

Article 17

1. Everyone has the right to own property alone as well as in association with others.

2. No one shall be arbitrarily deprived of his property.

[...]

Article 25

1. Everyone has the right to a standard of living adequate for the health and well-being of himself and of his family, including food, clothing, housing and medical care and necessary social services, and the right to security in the event of unemployment, sickness, disability, widowhood, old age or other lack of livelihood in circumstances beyond his control.

[...]

⁵ Adopted on 10 December 1948.

⁴ Adopted on 8 June and entered into force on 7 December 1978 in accordance with Article 23.

1.6. International Convention on the Elimination of All Forms of Racial Discrimination (1965)⁶

[...]

Article 5(e)(iii)

In compliance with the fundamental obligations laid down in article 2 of this Convention, States Parties undertake to prohibit and eliminate racial discrimination in all of its forms and to guarantee the right of everyone, without distinction as to race, colour, or national or ethnic origin, to equality before the law, notably in the enjoyment of the following rights: . . . (e) in particular . . . (iii) the right to housing.

[...]

Article 6

States Parties shall assure to everyone within their jurisdiction effective protection and remedies, through the competent national tribunals and other State institutions, against any acts of racial discrimination which violate his human rights and fundamental freedoms contrary to this Convention, as well as the right to seek from such tribunals just and adequate reparation or satisfaction for any damage suffered as a result of such discrimination.

[...]

1.7. International Covenant on Economic, Social and Cultural Rights (1966)⁷

[...]

Article 2

1. Each State Party to the present Covenant undertakes to take steps, individually and through international assistance and co-operation, especially economic and technical, to the maximum of its available resources, with a view to achieving progressively the full realization of the rights recognized in the present Covenant by all appropriate means, including particularly the adoption of legislative measures.

2. The States Parties to the present Covenant undertake to guarantee that the rights enunciated in the present Covenant will be exercised without discrimination of any kind as to race, colour, sex, language, religion, political or other opinion, national or social origin, property, birth or other status.

3. Developing countries, with due regard to human rights and their national economy, may determine to what extent they would guarantee the economic rights recognized in the present Covenant to non-nationals.

[...]

Article 11

1. The States Parties to the present Covenant recognize the right of everyone to an adequate standard of living for himself and his family, including adequate food, clothing and housing, and to the continuous improvement of living conditions. The States Parties will take appropriate steps to ensure the realization of this right, recognizing to this effect the essential importance of international co-operation based on free consent.

[...]

1.8. International Covenant on Civil and Political Rights (1966)⁸

[...]

Part II

Article 2

1. Each State Party to the present Covenant undertakes to respect and to ensure to all individuals within its territory and subject to its jurisdiction the rights recognized in the present Covenant, without distinction of any kind, such as race, colour, sex, language, religion, political or other opinion, national or social origin, property, birth or other status.

⁶ Adopted on 21 December 1965 and entered into force on 4 January 1969 in accordance with Article 19.

⁷ Adopted on 16 December 1966 and entered into force on 23 March 1976 in accordance with Article 27.

⁸ Adopted on 16 December 1966 and entered into force on 23 March 1976 in accordance with Article 49.

1.9. Convention on the Rights of the Child (1989)

2. Where not already provided for by existing legislative or other measures, each State Party to the present Covenant undertakes to take the necessary steps, in accordance with its constitutional processes and with the provisions of the present Covenant, to adopt such laws or other measures as may be necessary to give effect to the rights recognized in the present Covenant.

3. Each State Party to the present Covenant undertakes:

(a) To ensure that any person whose rights or freedoms as herein recognized are violated shall have an effective remedy, notwithstanding that the violation has been committed by persons acting in an official capacity;

(b) To ensure that any person claiming such a remedy shall have his right thereto determined by competent judicial, administrative or legislative authorities, or by any other competent authority provided for by the legal system of the State, and to develop the possibilities of judicial remedy;

(c) To ensure that the competent authorities shall enforce such remedies when granted.

[...]

PART III

Article 12

1. Everyone lawfully within the territory of a State shall, within that territory, have the right to liberty of movement and freedom to choose his residence.

2. Everyone shall be free to leave any country, including his own.

3. The above-mentioned rights shall not be subject to any restrictions except those which are provided by law, are necessary to protect national security, public order (ordre public), public health or morals or the rights and freedoms of others, and are consistent with the other rights recognized in the present Covenant.

4. No one shall be arbitrarily deprived of the right to enter his own country.

Article 17

1. No one shall be subjected to arbitrary or unlawful interference with his privacy, family, home or correspondence, nor to unlawful attacks on his honour and reputation.

2. Everyone has the right to the protection of the law against such interference or attacks.

[...]

1.9. Convention on the Rights of the Child (1989)⁹

[...]

Article 16

1. No child shall be subjected to arbitrary or unlawful interference with his or her privacy, family, home or correspondence, nor to unlawful attacks on his or her honour and reputation.

2. The child has the right to the protection of the law against such interference or attacks.

[...]

Article 22

1. States Parties shall take appropriate measures to ensure that a child who is seeking refugee status or who is considered a refugee in accordance with applicable international or domestic law and procedures shall, whether unaccompanied or accompanied by his or her parents or by any other person, receive appropriate protection and humanitarian assistance in the enjoyment of applicable rights set forth in the present Convention and in other international human rights or humanitarian instruments to which the said States are Parties.

2. For this purpose, States Parties shall provide, as they consider appropriate, co-operation in any efforts by the United Nations and other competent intergovernmental organizations or nongovernmental organizations co-operating with the United Nations to protect and assist such a child and to trace the parents or other members of the family of any refugee child in order to obtain

⁹ Adopted on 20 November 1989 and entered into force on 2 September 1990 in accordance with Article 49.

information necessary for reunification with his or her family. In cases where no parents or other members of the family can be found, the child shall be accorded the same protection as any other child permanently or temporarily deprived of his or her family environment for any reason, as set forth in the present Convention.

[...]

1.10. Convention (No. 169) Concerning Indigenous and Tribal Peoples in Independent Countries (1989)¹⁰

[...]

Article 16

1. Subject to the following paragraphs of this Article, the peoples concerned shall not be removed from the lands which they occupy.

2. Where the relocation of these peoples is considered necessary as an exceptional measure, such relocation shall take place only with their free and informed consent. Where their consent cannot be obtained, such relocation shall take place only following appropriate procedures established by national laws and regulations, including public inquiries where appropriate, which provide the opportunity for effective representation of the peoples concerned.

3. Whenever possible, these peoples shall have the right to return to their traditional lands, as soon as the grounds for relocation cease to exist.

4. When such return is not possible, as determined by agreement or, in the absence of such agreement, through appropriate procedures, these peoples shall be provided in all possible cases with lands of quality and legal status at least equal to that of the lands previously occupied by them, suitable to provide for their present needs and future development. Where the peoples concerned express a preference for compensation in money or in kind, they shall be so compensated under appropriate guarantees. 5. Persons thus relocated shall be fully compensated for any resulting loss or injury.

[...]

INTERNATIONAL CRIMINAL LAW

1.11. Rome Statute of the International Criminal Court (1998)¹¹

[...]

Article 5 – Crimes within the Jurisdiction of the Court

1. The jurisdiction of the Court shall be limited to the most serious crimes of concern to the international community as a whole. The Court has jurisdiction in accordance with this Statute with respect to the following crimes:

- (a) The crime of genocide;
- (b) Crimes against humanity;
- (c) War crimes;
- (d) The crime of aggression.

2. The Court shall exercise jurisdiction over the crime of aggression once a provision is adopted in accordance with articles 121 and 123 defining the crime and setting out the conditions under which the Court shall exercise jurisdiction with respect to this crime. Such a provision shall be consistent with the relevant provisions of the Charter of the United Nations.

Article 6 – Genocide

For the purpose of this Statute, "genocide" means any of the following acts committed with intent to destroy, in whole or in part, a national, ethnical, racial or religious group, as such:

(a) Killing members of the group;

(b) Causing serious bodily or mental harm to members of the group;

(c) Deliberately inflicting on the group conditions of life calculated to bring about its physical destruction in whole or in part;

(d) Imposing measures intended to prevent births within the group;

¹⁰ Adopted on 27 June 1989 and entered into force on 5 September 1991 in accordance with Article 38.

¹¹ Adopted on 17 July 1998 and entered into force on 1 July 2002 in accordance with Article 126.

(e) Forcibly transferring children of the group to another group.

Article 7 - Crimes against Humanity

1. For the purpose of this Statute, "crime against humanity" means any of the following acts when committed as part of a widespread or systematic attack directed against any civilian population, with knowledge of the attack:

- (a) Murder;
- (b) Extermination;
- (c) Enslavement;
- (d) Deportation or forcible transfer of population;

(e) Imprisonment or other severe deprivation of physical liberty in violation of fundamental rules of international law;

(f) Torture;

(g) Rape, sexual slavery, enforced prostitution, forced pregnancy, enforced sterilization, or any other form of sexual violence of comparable gravity;

(h) Persecution against any identifiable group or collectivity on political, racial, national, ethnic, cultural, religious, gender as defined in paragraph 3, or other grounds that are universally recognized as impermissible under international law, in connection with any act referred to in this paragraph or any crime within the jurisdiction of the Court;

- (i) Enforced disappearance of persons;
- (j) The crime of apartheid;

(k) Other inhumane acts of a similar character intentionally causing great suffering, or serious injury to body or to mental or physical health.

2. For the purpose of paragraph 1:

(a) "Attack directed against any civilian population" means a course of conduct involving the multiple commission of acts referred to in paragraph 1 against any civilian population, pursuant to or in furtherance of a State or organizational policy to commit such attack;

(b) "Extermination" includes the intentional infliction of conditions of life, inter alia the deprivation of access to food and medicine, calculated to bring about the destruction of part of a population; (c) "Enslavement" means the exercise of any or all of the powers attaching to the right of ownership over a person and includes the exercise of such power in the course of trafficking in persons, in particular women and children;

(d) "Deportation or forcible transfer of population" means forced displacement of the persons concerned by expulsion or other coercive acts from the area in which they are lawfully present, without grounds permitted under international law;

(e) "Torture" means the intentional infliction of severe pain or suffering, whether physical or mental, upon a person in the custody or under the control of the accused; except that torture shall not include pain or suffering arising only from, inherent in or incidental to, lawful sanctions;

(f) "Forced pregnancy" means the unlawful confinement of a woman forcibly made pregnant, with the intent of affecting the ethnic composition of any population or carrying out other grave violations of international law. This definition shall not in any way be interpreted as affecting national laws relating to pregnancy;

(g) "Persecution" means the intentional and severe deprivation of fundamental rights contrary to international law by reason of the identity of the group or collectivity;

(h) "The crime of apartheid" means inhumane acts of a character similar to those referred to in paragraph 1, committed in the context of an institutionalized regime of systematic oppression and domination by one racial group over any other racial group or groups and committed with the intention of maintaining that regime;

(i) "Enforced disappearance of persons" means the arrest, detention or abduction of persons by, or with the authorization, support or acquiescence of, a State or a political organization, followed by a refusal to acknowledge that deprivation of freedom or to give information on the fate or whereabouts of those persons, with the intention of removing them from the protection of the law for a prolonged period of time.

3. For the purpose of this Statute, it is understood that the term "gender" refers to the two sexes, male and female, within the context of society. The term "gender" does not indicate any meaning different from the above.

Article 8 - War Crimes

1. The Court shall have jurisdiction in respect of war crimes in particular when committed as part of a plan or policy or as part of a large-scale commission of such crimes.

2. For the purpose of this Statute, "war crimes" means:

(a) Grave breaches of the Geneva Conventions of 12 August 1949, namely, any of the following acts against persons or property protected under the provisions of the relevant Geneva Convention:

(i) Wilful killing;

(ii) Torture or inhuman treatment, including biological experiments;

(iii) Wilfully causing great suffering, or serious injury to body or health;

(iv) Extensive destruction and appropriation of property, not justified by military necessity and carried out unlawfully and wantonly;

(v) Compelling a prisoner of war or other protected person to serve in the forces of a hostile Power;

(vi) Wilfully depriving a prisoner of war or other protected person of the rights of fair and regular trial;

(vii) Unlawful deportation or transfer or unlawful confinement;

(viii) Taking of hostages.

(b) Other serious violations of the laws and customs applicable in international armed conflict, within the established framework of international law, namely, any of the following acts:

(i) Intentionally directing attacks against the civilian population as such or against individual civilians not taking direct part in hostilities;

(ii) Intentionally directing attacks against civilian objects, that is, objects which are not military objectives;

(iii) Intentionally directing attacks against personnel, installations, material, units or vehicles involved in a humanitarian assistance or peacekeeping mission in accordance with the Charter of the United Nations, as long as they are entitled to the protection given to civilians or civilian objects under the international law of armed conflict; (iv) Intentionally launching an attack in the knowledge that such attack will cause incidental loss of life or injury to civilians or damage to civilian objects or widespread, long-term and severe damage to the natural environment which would be clearly excessive in relation to the concrete and direct overall military advantage anticipated;

(v) Attacking or bombarding, by whatever means, towns, villages, dwellings or buildings which are undefended and which are not military objectives;

(vi) Killing or wounding a combatant who, having laid down his arms or having no longer means of defence, has surrendered at discretion;

(vii) Making improper use of a flag of truce, of the flag or of the military insignia and uniform of the enemy or of the United Nations, as well as of the distinctive emblems of the Geneva Conventions, resulting in death or serious personal injury;

(viii) The transfer, directly or indirectly, by the Occupying Power of parts of its own civilian population into the territory it occupies, or the deportation or transfer of all or parts of the population of the occupied territory within or outside this territory;

(ix) Intentionally directing attacks against buildings dedicated to religion, education, art, science or charitable purposes, historic monuments, hospitals and places where the sick and wounded are collected, provided they are not military objectives;

(x) Subjecting persons who are in the power of an adverse party to physical mutilation or to medical or scientific experiments of any kind which are neither justified by the medical, dental or hospital treatment of the person concerned nor carried out in his or her interest, and which cause death to or seriously endanger the health of such person or persons;

(xi) Killing or wounding treacherously individuals belonging to the hostile nation or army;

(xii) Declaring that no quarter will be given;

(xiii) Destroying or seizing the enemy's property unless such destruction or seizure be imperatively demanded by the necessities of war;

(xiv) Declaring abolished, suspended or inadmissible in a court of law the rights and actions of the nationals of the hostile party;

1.11. Rome Statute of the International Criminal Court (1998)

(xv) Compelling the nationals of the hostile party to take part in the operations of war directed against their own country, even if they were in the belligerent's service before the commencement of the war;

(xvi) Pillaging a town or place, even when taken by assault;

(xvii) Employing poison or poisoned weapons;

(xviii) Employing asphyxiating, poisonous or other gases, and all analogous liquids, materials or devices;

(xix) Employing bullets which expand or flatten easily in the human body, such as bullets with a hard envelope which does not entirely cover the core or is pierced with incisions;

(xx) Employing weapons, projectiles and material and methods of warfare which are of a nature to cause superfluous injury or unnecessary suffering or which are inherently indiscriminate in violation of the international law of armed conflict, provided that such weapons, projectiles and material and methods of warfare are the subject of a comprehensive prohibition and are included in an annex to this Statute, by an amendment in accordance with the relevant provisions set forth in articles 121 and 123;

(xxi) Committing outrages upon personal dignity, in particular humiliating and degrading treatment;

(xxii) Committing rape, sexual slavery, enforced prostitution, forced pregnancy, as defined in article 7, paragraph 2 (f), enforced sterilization, or any other form of sexual violence also constituting a grave breach of the Geneva Conventions;

(xxiii) Utilizing the presence of a civilian or other protected person to render certain points, areas or military forces immune from military operations;

(xxiv) Intentionally directing attacks against buildings, material, medical units and transport, and personnel using the distinctive emblems of the Geneva Conventions in conformity with international law;

(xxv) Intentionally using starvation of civilians as a method of warfare by depriving them of objects indispensable to their survival, including wilfully impeding relief supplies as provided for under the Geneva Conventions; (xxvi) Conscripting or enlisting children under the age of fifteen years into the national armed forces or using them to participate actively in hostilities.

(c) In the case of an armed conflict not of an international character, serious violations of article 3 common to the four Geneva Conventions of 12 August 1949, namely, any of the following acts committed against persons taking no active part in the hostilities, including members of armed forces who have laid down their arms and those placed hors de combat by sickness, wounds, detention or any other cause:

(i) Violence to life and person, in particular murder of all kinds, mutilation, cruel treatment and torture;

(ii) Committing outrages upon personal dignity, in particular humiliating and degrading treatment;

(iii) Taking of hostages;

(iv) The passing of sentences and the carrying out of executions without previous judgement pronounced by a regularly constituted court, affording all judicial guarantees which are generally recognized as indispensable.

(d) Paragraph 2 (c) applies to armed conflicts not of an international character and thus does not apply to situations of internal disturbances and tensions, such as riots, isolated and sporadic acts of violence or other acts of a similar nature.

(e) Other serious violations of the laws and customs applicable in armed conflicts not of an international character, within the established framework of international law, namely, any of the following acts:

(i) Intentionally directing attacks against the civilian population as such or against individual civilians not taking direct part in hostilities;

(ii) Intentionally directing attacks against buildings, material, medical units and transport, and personnel using the distinctive emblems of the Geneva Conventions in conformity with international law;

(iii) Intentionally directing attacks against personnel, installations, material, units or vehicles involved in a humanitarian assistance or peacekeeping mission in accordance with the Charter of the United Nations, as long as they are entitled to the protection given to civilians or civilian objects under the international law of armed conflict; (iv) Intentionally directing attacks against buildings dedicated to religion, education, art, science or charitable purposes, historic monuments, hospitals and places where the sick and wounded are collected, provided they are not military objectives;

(v) Pillaging a town or place, even when taken by assault;

(vi) Committing rape, sexual slavery, enforced prostitution, forced pregnancy, as defined in article 7, paragraph 2 (f), enforced sterilization, and any other form of sexual violence also constituting a serious violation of article 3 common to the four Geneva Conventions;

(vii) Conscripting or enlisting children under the age of fifteen years into armed forces or groups or using them to participate actively in hostilities;

(viii) Ordering the displacement of the civilian population for reasons related to the conflict, unless the security of the civilians involved or imperative military reasons so demand;

(ix) Killing or wounding treacherously a combatant adversary;

(x) Declaring that no quarter will be given;

(xi) Subjecting persons who are in the power of another party to the conflict to physical mutilation or to medical or scientific experiments of any kind which are neither justified by the medical, dental or hospital treatment of the person concerned nor carried out in his or her interest, and which cause death to or seriously endanger the health of such person or persons;

(xii) Destroying or seizing the property of an adversary unless such destruction or seizure be imperatively demanded by the necessities of the conflict;

(f) Paragraph 2 (e) applies to armed conflicts not of an international character and thus does not apply to situations of internal disturbances and tensions, such as riots, isolated and sporadic acts of violence or other acts of a similar nature. It applies to armed conflicts that take place in the territory of a State when there is protracted armed conflict between governmental authorities and organized armed groups or between such groups. 3. Nothing in paragraph 2 (c) and (e) shall affect the responsibility of a Government to maintain or re-establish law and order in the State or to defend the unity and territorial integrity of the State, by all legitimate means.

[...]

Article 75 - Reparation to Victims

1. The Court shall establish principles relating to reparations to, or in respect of, victims, including restitution, compensation and rehabilitation. On this basis, in its decision the Court may, either upon request or on its own motion in exceptional circumstances, determine the scope and extent of any damage, loss and injury to, or in respect of, victims and will state the principles on which it is acting.

2. The Court may make an order directly against a convicted person specifying appropriate reparations to, or in respect of, victims, including restitution, compensation and rehabilitation.

Where appropriate, the Court may order that the award for reparations be made through the Trust Fund provided for in article 79.

3. Before making an order under this article, the Court may invite and shall take account of representations from or on behalf of the convicted person, victims, other interested persons or interested States.

4. In exercising its power under this article, the Court may, after a person is convicted of a crime within the jurisdiction of the Court, determine whether, in order to give effect to an order which it may make under this article, it is necessary to seek measures under article 93, paragraph 1.

5. A State Party shall give effect to a decision under this article as if the provisions of article 109 were applicable to this article.

6. Nothing in this article shall be interpreted as prejudicing the rights of victims under national or international law.

INTERNATIONAL REFUGEE LAW

1.12. Convention Relating to the Status of Refugees (1951)¹²

[...]

Article 21 - Housing

As regards housing, the Contracting States, in so far as the matter is regulated by laws or regulations or is subject to the control of public authorities, shall accord to refugees lawfully staying in their territory treatment as favourable as possible and, in any event, not less favourable than that accorded to aliens generally in the same circumstances.

[...]

Article 26 – Freedom of Movement

Each Contracting State shall accord to refugees lawfully in its territory the right to choose their place of residence to move freely within its territory, subject to any regulations applicable to aliens generally in the same circumstances.

[...]

1.13. Protocol Relating to the Status of Refugees (1967)¹³

The States Parties to the present Protocol,

Considering that the Convention relating to the Status of Refugees done at Geneva on 28 July 1951 (hereinafter referred to as the Convention) covers only those persons who have become refugees as a result of events occurring before 1 January 1951,

Considering that new refugee situations have arisen since the Convention was adopted and that the refugees concerned may therefore not fall within the scope of the Convention,

Considering that it is desirable that equal status should be enjoyed by all refugees covered by the definition in the Convention irrespective of the dateline 1 January 1951,

Have agreed as follows:

Article 1 - General Provisions

1. The States Parties to the present Protocol undertake to apply articles 2 to 34 inclusive of the Convention to refugees as hereinafter defined.

2. For the purpose of the present Protocol, the term "refugee" shall, except as regards the application of paragraph 3 of this article, mean any person within the definition of article I of the Convention as if the words "As a result of events occurring before 1 January 1951 and ..." and the words "... as a result of such events", in article 1 A (2) were omitted.

3. The present Protocol shall be applied by the States Parties hereto without any geographic limitation, save that existing declarations made by States already Parties to the Convention in accordance with article I B (I) (a) of the Convention, shall, unless extended under article I B (2) thereof, apply also under the present Protocol.

Article 2 – Co-Operation of the National Authorities with the United Nations

1. The States Parties to the present Protocol undertake to co-operate with the Office of the United Nations High Commissioner for Refugees, or any other agency of the United Nations which may succeed it, in the exercise of its functions, and shall in particular facilitate its duty of supervising the application of the provisions of the present Protocol.

2. In order to enable the Office of the High Commissioner or any other agency of the United Nations which may succeed it, to make reports to the competent organs of the United Nations, the States Parties to the present Protocol undertake to provide them with the information and statistical data requested, in the appropriate form, concerning:

- (a) The condition of refugees;
- (b) The implementation of the present Protocol;

(c) Laws, regulations and decrees which are, or may hereafter be, in force relating to refugees.

Article 3 - Information on National Legislation

The States Parties to the present Protocol shall communicate to the Secretary-General of the United Nations the laws and regulations which they may

¹² Adopted on 28 July 1951 and entered into force on 22 April 1954, in accordance with Article 43.

¹³ Adopted on 16 December 1966 and entered into force on 4 October 1967 in accordance with Article 8.

[...]

THE INTERNATIONAL LAW OF STATE RESPONSIBILITY

1.14. International Law Commission's Draft Articles on Responsibility of States for Internationally Wrongful Acts (2001)¹⁴

[...]

Article 3

The characterization of an act of a State as internationally wrongful is governed by international law. Such characterization in not affected by the characterization of the same act lawful by internal law.

[...]

Article 14(2)

The breach of an international obligation by an act of a State having a continuing character extends over the entire period during which the act continues and remains not in conformity with the international obligation.

[...]

Article 29

The legal consequences of an internationally wrongful act under the Part do not affect the continued duty of the responsible State to perform the obligation breached.

Article 30

The State responsible for the internationally wrongful act is under an obligation: (b) to offer appropriate assurances and guarantees of nonrepetition, if circumstances so require.

Article 31

The responsible State may not rely on the provisions of its internal law as justification for failure to comply with its obligations. [...]

Reparation for Injury

Article 34 - Forms of Reparation

Full reparation for the injury caused by the internationally wrongful act shall take the form of restitution, compensation and satisfaction, either singly or in combination, in accordance with the provisions of this chapter.

Article 35 – Restitution

A State responsible for an internationally wrongful act is under an obligation to make restitution, that is, to re-establish the situation which existed before the wrongful act was committed, provided and to the extent that restitution:

(a) Is not materially impossible;

(b) Does not involve a burden out of all proportion to the benefit deriving from restitution instead of compensation.

Article 36 - Compensation

1. The State responsible for an internationally wrongful act is under an obligation to compensate for the damage caused thereby, insofar as such damage is not made good by restitution.

2. The compensation shall cover any financially assessable damage including loss of profits insofar as it is established.

Article 37 - Satisfaction

1. The State responsible for an internationally wrongful act is under an obligation to give satisfaction for the injury caused by that act insofar as it cannot be made good by restitution or compensation.

2. Satisfaction may consist in an acknowledgement of the breach, an expression of regret, a formal apology or another appropriate modality.

3. Satisfaction shall not be out of proportion to the injury and may not take a form humiliating to the responsible State.

Article 38

1. The State responsible for an internationally wrongful act is under an obligation to give satisfaction for the injury caused by that act insofar as

¹⁴ Fifty-Third Session (2001) Extract from the Report of the International Law Commission on the work of its Fifty-Third Session, Official Records of the General Assembly, Fifty-Sixth Session, Supplement No. 10 (A/56/10), ch.IV.E.1.

it cannot be made good by restitution or compensation.

2. Satisfaction may consist in an acknowledgement of the breach, an expression of regret, a formal apology or another appropriate modality.

3. Satisfaction shall not be out of proportion to the injury and may not take a form humiliating to the responsible State.

[...]

SPECIFIC RESTITUTION STANDARDS

1.15. 'Pinheiro' Principles on Housing and Property Restitution for Refugees and Displaced Persons (2005)¹⁵

Preamble

Recognizing that millions of refugees and displaced persons worldwide continue to live in precarious and uncertain situations, and that all refugees and displaced persons have a right to voluntary return, in safety and dignity, to their original or former habitual homes and lands;

Underscoring that voluntary return in safety and dignity must be based on a free, informed, individual choice and that refugees and displaced persons should be provided with complete, objective, up to date, and accurate information, including on physical, material and legal safety issues in countries or places of origin;

Reaffirming the rights of refugee and displaced women and girls, and recognizing the need to undertake positive measures to ensure that their rights to housing, land and property restitution are guaranteed;

Welcoming the many national and international institutions that have been established in recent years to ensure the restitution rights of refugees and displaced persons, as well as the many national and international laws, standards, policy statements, agreements and guidelines that have recognized and reaffirmed the right to housing, land and property restitution;

Convinced that the right to housing, land and property restitution is essential to the resolution of conflict and to post-conflict peace-building, safe and sustainable return and the establishment of the rule of law, and that careful monitoring of restitution Programs, on the part of international organizations and affected states, is indispensable to ensuring their effective implementation;

Convinced also that the implementation of successful housing, land and property restitution Programs, as a key element of restorative justice, contributes to effectively deterring future situations of displacement and building sustainable peace;

SECTION I. SCOPE AND APPLICATION

1. Scope and Application

1.1 The Principles on Housing and Property Restitution for Refugees and Displaced Persons articulated herein are designed to assist all relevant actors, national and international, in addressing the legal and technical issues surrounding housing, land and property restitution in situations where displacement has led to persons being arbitrarily or unlawfully deprived of their former homes, lands, properties or places of habitual residence.

1.2 The Principles on Housing and Property Restitution for Refugees and Displaced Persons apply equally to all refugees, internally displaced persons and to other similarly situated displaced persons who fled across national borders but who may not meet the legal definition of refugee, (hereinafter 'refugees and displaced persons') who were arbitrarily or unlawfully deprived of their former homes, lands, properties or places of habitual residence, regardless of the nature or circumstances by which displacement originally occurred.

Section II. The Right to Housing and Property Restitution

2. The Right to Housing and Property Restitution

2.1 All refugees and displaced persons have the right to have restored to them any housing, land and/or property of which they were arbitrarily or unlawfully deprived, or to be compensated for any housing, land and/or property that is factually

¹⁵ Adopted without a vote on 11 August 2005 in Resolution 2005/21 by the UN Sub-Commission on Protection and Promotion of Human Rights, Geneva.

impossible to restore as determined by an independent, impartial tribunal.

2.2 States shall demonstrably prioritize the right to restitution as the preferred remedy to displacement and as a key element of restorative justice. The right to restitution exists as a distinct right, and is prejudiced neither by the actual return nor nonreturn of refugees and displaced persons entitled to housing, land and property restitution.

SECTION III. OVERARCHING PRINCIPLES

3. The Right to Non-Discrimination

3.1 Everyone has the right to non-discrimination on the basis of race, color, sex, language, religion, political or other opinion, national or social origin, property, birth or other status.

3.2 States shall ensure that *de facto* and *de jure* discrimination on the above grounds is prohibited and that all persons, including refugees and displaced persons, are considered equal before the law.

4. The Right to Equality between Men and Women

4.1 States shall ensure the equal right of men and women, and the equal right of boys and girls, to the enjoyment of housing, land and property restitution. In particular, States shall ensure the equal right of men and women, and the equal right of boys and girls, to *inter alia* voluntary return in safety and dignity; legal security of tenure; property ownership; equal access to inheritance; as well as the use, control of and access to housing, land and property.

4.2 States should ensure that housing, land and property restitution Programs, policies and practices recognize the joint ownership rights of both the male and female heads of the household as an explicit component of the restitution process, and that restitution Programs, policies and practices reflect a gender sensitive approach.

4.3 States shall ensure that housing, land and property restitution Programs, policies and practices do not disadvantage women and girls. States should adopt positive measures to ensure gender equality in this regard.

5. The Right to Be Protected from Displacement

5.1 Everyone has the right to be protected against being arbitrarily displaced from his or her home, land or place of habitual residence.

5.2 States should incorporate protections against displacement into domestic legislation, consistent with international human rights and humanitarian law and related standards, and should extend these protections to everyone within their legal jurisdiction or effective control.

5.3 States shall prohibit forced eviction, demolition of houses and destruction of agricultural areas and the arbitrary confiscation or expropriation of land as a punitive measure or as a means or method of war.

5.4 States shall take steps to ensure that no one is subjected to displacement by either State or non-State actors. States shall also ensure that individuals, corporations, and other entities within their legal jurisdiction or effective control refrain from carrying out or otherwise participating in displacement.

6. The Right to Privacy and Respect for the Home

6.1 Everyone has the right to be protected against arbitrary or unlawful interference with his or her privacy and his or her home.

6.2 States shall ensure that everyone is provided with safeguards of due process against such arbitrary or unlawful interference with his or her privacy and his or her home.

7. The Right to Peaceful Enjoyment of Possessions

7.1 Everyone has the right to the peaceful enjoyment of his or her possessions.

7.2 States shall only subordinate the use and enjoyment of possessions in the public interest and subject to the conditions provided for by law and by the general Principles of international law. Whenever possible, the 'interest of society' should be read restrictively, so as to mean only a temporary interference with the right to peaceful enjoyment of possessions.

8. The Right to Adequate Housing

8.1 Everyone has the right to adequate housing.

8.2 States should adopt positive measures aimed at alleviating the situation of refugees and displaced persons living in inadequate housing.

9. The Right to Freedom of Movement

9.1 Everyone has the right to freedom of movement and the right to choose his or her residence. No one shall be arbitrarily or unlawfully forced to remain within a certain territory, area or region. Similarly, no one shall be arbitrarily or unlawfully forced to leave a certain territory, area or region.

9.2 States shall ensure that freedom of movement and the right to choose one's residence are not subject to any restrictions except those which are provided by law, are necessary to protect national security, public order, public health or morals or the rights and freedoms of others, and are consistent with international human rights, refugee and humanitarian law and related standards.

Section IV. The Right to Voluntary Return in Safety and Dignity

10. The Right to Voluntary Return in Safety and Dignity

10.1 All refugees and displaced persons have the right to voluntarily return to their former homes, lands or places of habitual residence, in safety and dignity. Voluntary return in safety and dignity must be based on a free, informed, individual choice. Refugees and displaced persons should be provided with complete, objective, up to date, and accurate information, including on physical, material and legal safety issues in countries or places of origin.

10.2 States shall allow refugees and displaced persons who wish to return voluntarily to their former homes, lands or places of habitual residence to do so. This right cannot be abridged under conditions of state succession, nor can it be subject to arbitrary or unlawful time limitations.

10.3 Refugees and displaced persons shall not be forced, or otherwise coerced, either directly or indirectly, to return to their former homes, lands or places of habitual residence. Refugees and displaced persons should be able to effectively pursue durable solutions to displacement other than return, if they so wish, without prejudicing their right to the restitution of their housing, land and property. 10.3 States should, when necessary, request from other States or international organizations the financial and/or technical assistance required to facilitate the effective voluntary return, in safety and dignity, of refugees and displaced persons.

Section V. Legal, Policy, Procedural and Institutional Implementation Mechanisms

11. Compatibility with International Human Rights, Refugee and Humanitarian Law and Related Standards

11.1 States should ensure that all housing, land and property restitution procedures, institutions, mechanisms and legal frameworks are fully compatible with international human rights, refugee and humanitarian law and related standards, and that the right to voluntary return in safety and dignity is recognized therein.

12. National Procedures, Institutions and Mechanisms

12.1 States should establish and support equitable, timely, independent, transparent and nondiscriminatory procedures, institutions and mechanisms to assess and enforce housing, land and property restitution claims. In cases where existing procedures, institutions and mechanisms can effectively address these issues, adequate financial, human and other resources should be made available to facilitate restitution in a just and timely manner.

12.2 States should ensure that housing, land and property restitution procedures, institutions and mechanisms are age and gender sensitive, and recognize the equal rights of men and women, as well as the equal rights of boys and girls, and reflect the overarching principle of the "best interests" of the child.

12.3 States should take all appropriate administrative, legislative and judicial measures to support and facilitate the housing, land and property restitution process. States should provide all relevant agencies with adequate financial, human and other resources to successfully complete their work in a just and timely manner.

12.4 States should establish guidelines which ensure the effectiveness of all relevant housing, land and property restitution procedures, institutions and mechanisms, including guidelines pertaining to institutional organization, staff training and caseloads, investigation and complaints procedures, verification of property ownership or other possessory rights, as well as decision-making, enforcement and appeals mechanisms. States may integrate alternative or informal dispute resolution mechanisms into these processes, insofar as all such mechanisms act in accordance with international human rights, refugee and humanitarian law and related standards, including the right to non-discrimination.

12.5 States should, where there has been a general breakdown in the rule of law, or where States are unable to implement the procedures, institutions and mechanisms necessary to facilitate the housing, land and property restitution process in a just and timely manner, request the technical assistance and cooperation of relevant international agencies in order to establish provisional regimes responsible for providing refugees and displaced persons with the procedures, institutions and mechanisms necessary to ensure effective restitution remedies.

12.6 States should include housing, land and property restitution procedures, institutions and mechanisms in peace agreements and voluntary repatriation agreements. Peace agreements should include specific undertakings by the parties to appropriately address any housing, land and property issues that require remedies under international law or threaten to undermine the peace process if left unaddressed, while demonstrably prioritizing the right to restitution as the preferred remedy in this regard.

13. Accessibility of Restitution Claims Procedures

13.1 Everyone who has been arbitrarily or unlawfully deprived of housing, land and/or property should be able to submit a claim for restitution and/or compensation to an independent and impartial body, and to receive a determination on their claim. States should not establish any preconditions for filing a restitution claim.

13.2 States should ensure that all aspects of the restitution claims process, including appeals procedures, are just, timely, accessible, free of charge, and are age and gender sensitive. States should adopt positive measures to ensure that women are

able to participate on a fully equal basis in this process.

13.3 States should ensure that separated and unaccompanied children are able to participate and are fully represented in the restitution claims process, and that any decision in relation to the restitution claim of separated and unaccompanied children is in compliance with the overarching principle of the "best interests" of the child.

13.4 States should ensure that the restitution claims process is accessible for refugees and other displaced persons regardless of their place of residence during the period of displacement, including in countries of origin, countries of asylum or countries to which they have fled. States should ensure that all affected persons are made aware of the restitution claims process, and that information about this process is made readily available, including in countries of origin, countries of asylum or countries to which they have fled.

13.5 States should seek to establish restitution claims processing centers and offices throughout affected areas where potential claimants currently reside. In order to facilitate the greatest access to those affected, it should be possible to submit restitution claims by post or by proxy, as well as in person. States should also consider establishing mobile units in order to ensure accessibility to all potential claimants.

13.6 States should ensure that users of housing, land and/or property, including tenants, have the right to participate in the restitution claims process, including through the filing of collective restitution claims.

13.7 States should develop restitution claims forms that are simple, easy to understand and use and make them available in the first language or languages of the groups affected. Competent assistance should be made available to help persons in completing and filing any necessary restitution claims forms, and such assistance should be provided in a manner which is age and gender sensitive.

13.8 Where restitution claims forms cannot be sufficiently simplified due to the complexities inherent in the claims process, States should engage qualified persons to interview potential claimants in confidence, and in a manner which is age and gender sensitive, in order to solicit the necessary information and complete the restitution claims forms on their behalf.

13.9 States should establish a clear time period for filing restitution claims. The time period should be widely disseminated and should be sufficiently long to ensure that all those affected have an adequate opportunity to file a restitution claim, bearing in mind the number of potential claimants, potential difficulties of information and access, the spread of displacement, the accessibility of the process for potentially disadvantaged groups and vulnerable individuals, and the political situation in the country or region of origin.

13.10 States should ensure that persons needing special assistance, including illiterate and disabled persons, are provided with such assistance in order to ensure that they are not denied access to the restitution claims process.

13.11 States should ensure that adequate legal aid is provided, if possible free of charge, to those seeking to make a restitution claim. While legal aid may be provided by either governmental or nongovernmental sources (be they national or international), such legal aid should meet adequate standards of quality, non-discrimination, fairness and impartiality so as not to prejudice the restitution claims process.

13.12 States should ensure that no one is persecuted or punished for making a restitution claim.

14. Adequate Consultation and Participation in Decision-Making

14.1 States and other involved international and national actors should ensure that voluntary repatriation and housing, land and property restitution Programs are carried out with adequate consultation and participation with the affected persons, groups and communities.

14.2 States and other involved international and national actors should, in particular, ensure that women, indigenous peoples, racial and ethnic minorities, the elderly, the disabled and children are adequately represented and included in restitution decision-making processes, and have the appropriate means and information to participate effectively. The needs of vulnerable individuals including the elderly, single female heads of households,

separated and unaccompanied children, and the disabled should be given particular attention.

15. Housing, Land and Property Records and Documentation

15.1 States should establish or re-establish national multi-purpose cadastre or other appropriate systems for the registration of housing, land and property rights as an integral component of any restitution Programs, respecting the rights of refugees and displaced persons when doing so.

15.2 States should ensure that any judicial, quasijudicial, administrative or customary pronouncement regarding the rightful ownership of, or rights to, housing, land and/or property is accompanied by measures to ensure registration or demarcation of that housing, land and/or property right as is necessary to ensure legal security of tenure. These determinations shall comply with international human rights, refugee and humanitarian law and related standards, including the right to nondiscrimination.

15.3 States should ensure, where appropriate, that registration systems record and/or recognize the possessory rights of traditional and indigenous communities to collective lands.

15.4 States and other responsible authorities or institutions should ensure that existing registration systems are not destroyed in times of conflict or post-conflict. Measures to prevent the destruction of housing, land and property records could include protection *in situ* or, if necessary, short-term removal to a safe location or custody. If removed, the records should be returned as soon as possible after the end of hostilities. States and other responsible authorities may also consider establishing procedures for copying records (including in digital format) transferring them securely, and recognizing the authenticity of said copies.

15.5 States and other responsible authorities or institutions should provide, at the request of a claimant or his or her proxy, copies of any documentary evidence in their possession required to make and/or support a restitution claim. Such documentary evidence should be provided free of charge, or for a minimal fee.

15.6 States and other responsible authorities or institutions conducting the registration of refugees or displaced persons should endeavor to collect

information relevant to facilitating the restitution process, for example by including in the registration form questions regarding the location and status of the individual refugee's or displaced person's former home, land, property or place of habitual residence. Such information should be sought whenever information is gathered from refugees and displaced persons, including at the time of flight.

15.7 States may, in situations of mass displacement where little documentary evidence exists as to ownership or possessory rights, adopt the conclusive presumption that persons fleeing their homes during a given period marked by violence or disaster have done so for reasons related to violence or disaster and are therefore entitled to housing, land and property restitution. In such cases, administrative and judicial authorities may independently establish the facts related to undocumented restitution claims.

15.8 States shall not recognize as valid any housing, land and/or property transaction, including any transfer that was made under duress, or which was otherwise coerced or forced, either directly or indirectly, or which was carried out contrary to international human rights standards.

16. The Rights of Tenants and Other Non-Owners

16.1 States should ensure that the rights of tenants, social occupancy rights holders and other legitimate occupants or users of housing, land and property are recognized within restitution Programs. To the maximum extent possible, States should ensure that such persons are able to return to and re-possess and use their housing, land and property in a similar manner to those possessing formal ownership rights.

17. Secondary Occupants

17.1 States should ensure that secondary occupants are protected against arbitrary or unlawful forced eviction. States shall ensure, in cases where evictions of such occupants are deemed justifiable and unavoidable for the purposes of housing, land and property restitution, that evictions are carried out in a manner which is compatible with international human rights law and standards, such that secondary occupants are afforded safeguards of due process, including, *inter alia*, an opportunity for genuine consultation, adequate and reasonable

notice, and the provision of legal remedies, including opportunities for legal redress.

17.2 States should ensure that the safeguards of due process extended to secondary occupants do not prejudice the rights of legitimate owners, tenants and other rights holders to repossess the housing, land and property in question in a just and timely manner.

17.3 States should, in cases where evictions of secondary occupants are justifiable and unavoidable, take positive measures to protect those who do not have the means to access any other adequate housing other than that which they are currently occupying from homelessness and other violations of their right to adequate housing. States should undertake to identify and provide alternative housing and/or land for such occupants, including on a temporary basis, as a means to facilitate the timely restitution of refugee and displaced persons housing, land and property. Lack of such alternatives, however, should not unnecessarily delay the implementation and enforcement of decisions by relevant bodies regarding housing, land and property restitution.

17.4 States may consider, in cases where housing, land and property has been sold by secondary occupants to third parties acting in good faith, establishing mechanisms to provide compensation to injured third parties. The egregiousness of the underlying displacement, however, may arguably give rise to constructive notice of the illegality of purchasing abandoned property, preempting the formation of *bona fide* property interests in such cases.

18. Legislative Measures

18.1 States should ensure the right of refugees and displaced persons to housing, land and property restitution is recognized as an essential component of the rule of law. States should ensure the right to housing, land and property restitution through all necessary legislative means, including through the adoption, amendment, reform, or repeal of relevant laws, regulations and/or practices. States should develop a legal framework for protecting the right to housing, land and property restitution which is clear, consistent and, where necessary, consolidated in a single law.

18.2 States should ensure that all relevant laws clearly delineate every person and/or affected group that is legally entitled to the restitution of their housing, land and property, most notably refugees and displaced persons. Subsidiary claimants should similarly be recognized, including resident family members at the time of displacement, spouses, domestic partners, dependents, legal heirs and others who should be entitled to claim on the same basis as primary claimants.

18.3 States should ensure that national legislation related to housing, land and property restitution is internally consistent, as well as compatible with pre-existing relevant agreements, such as peace agreements and voluntary repatriation agreements, so long as these agreements are themselves compatible with international human rights, refugee and humanitarian law and related standards.

19. Prohibition of Arbitrary and Discriminatory Laws

19.1 States should neither adopt nor apply laws which prejudice the restitution process, in particular through arbitrary, discriminatory, or otherwise unjust abandonment laws or statues of limitations.

9.2 States should take immediate steps to repeal unjust or arbitrary laws, and laws which otherwise have a discriminatory effect on the enjoyment of the right to housing, land and property restitution, and should ensure remedies for those wrongfully harmed by the prior application of such laws.

19.3 States should ensure that all national policies related to the right to housing, land and property restitution fully guarantee the rights of women and girls to non-discrimination and to equality in both law and practice.

20. Enforcement of Restitution Decisions and Judgments

20.1 States should designate specific public agencies to be entrusted with enforcing housing, land and property restitution decisions and judgments.

20.2 States should ensure, through law and other appropriate means, that local and national authorities are legally obligated to respect, implement and enforce decisions and judgments made by relevant bodies regarding housing, land and property restitution. 20.3 States should adopt specific measures to prevent the public obstruction of enforcement of housing, land and property restitution decisions and judgments. Threats or attacks against officials and agencies carrying out restitution Programs should be fully investigated and prosecuted.

20.4 States should adopt specific measures to prevent the destruction or looting of contested or abandoned housing, land and property. In order to minimize destruction and looting, States should develop procedures to inventory the contents of claimed housing, land and property within the context of housing, land and property restitution Programs.

20.5 States should implement public information campaigns aimed at informing secondary occupants and other relevant parties of their rights and of the legal consequences of non-compliance with housing, land and property restitution decisions and judgments, including failing to vacate occupied housing, land and property voluntarily and damaging and/or looting of occupied housing, land and property.

21. Compensation

21.1 All refugees and displaced persons have the right to full and effective compensation as an integral component of the restitution process. Compensation may be monetary or in kind. States shall, in order to comply with the principle of restorative justice, ensure that the remedy of compensation is only be used when the remedy of restitution is not factually possible or when the injured party knowingly and voluntarily accepts compensation in lieu of restitution, or when the terms of a negotiated peace settlement provide for a combination of restitution and compensation.

21.2 States should ensure, as a rule, that restitution is only deemed factually impossible in exceptional circumstances, namely when housing, land and/or property is destroyed or when it no longer exists, as determined by an independent, impartial tribunal. Even under such circumstances the holder of the housing, land and/or property right should have the option to repair or rebuild whenever possible. In some situations, a combination of compensation and restitution may be the most appropriate remedy and form of restorative justice.

Section VI. The Role of the International Community, Including International Organizations

22. Responsibility of the International Community

22.1 The international community should promote and protect the right to housing, land and property restitution, as well as the right to voluntary return in safety and dignity.

22.2 International financial, trade, development and other related institutions and agencies, including member or donor States that have voting rights within such bodies, should take fully into account the prohibition against unlawful or arbitrary displacement and, in particular, the prohibition under international human rights law and related standards on the practice of forced evictions.

22.3 International organizations should work with national governments and share expertise on the development of national housing, land and property restitution policies and Programs and help ensure their compatibility with international human rights, refugee and humanitarian law and related standards. International organizations should also support the monitoring of their implementation.

22.4 International organizations, including the United Nations, should strive to ensure that peace agreements and voluntary repatriation agreements contain provisions related to housing, land and property restitution, including through *inter alia* the establishment of national procedures, institutions, mechanisms and legal frameworks.

22.5 International peace operations, in pursuing their overall mandate, should help to maintain a secure and stable environment wherein appropriate housing, land and property restitution policies and Programs may be successfully implemented and enforced.

22.6 International peace operations, depending on the mission context, should be requested to support the protection of the right to housing, land and property restitution, including through the enforcement of restitution decisions and judgments. Member States in the Security Council should consider including this role in the mandate of peace operations. 22.7 International organizations and peace operations should avoid occupying, renting or purchasing housing, land and property over which the rights holder does not currently have access or control, and should require that their staff do the same. Similarly, international organizations and peace operations should ensure that bodies or processes under their control or supervision do not obstruct, directly or indirectly, the restitution of housing, land and property.

SECTION VII. INTERPRETATION

23. Interpretation

23.1 The *Principles on Housing and Property Restitution for Refugees and Displaced Persons* shall not be interpreted as limiting, altering or otherwise prejudicing the rights recognized under international human rights, refugee and humanitarian law and related standards, or rights consistent with these laws and standards as recognized under national law.

1.16. Basic Principles and Guidelines on the Right to a Remedy and Reparation for Victims of Violations of International Human Rights and Humanitarian Law (2005)¹⁶

Preamble

The General Assembly,

Recalling the provisions providing a right to a remedy for victims of violations of international human rights law found in numerous international instruments, in particular article 8 of the Universal Declaration of Human Rights, 1 article 2 of the International Covenant on Civil and Political Rights, 2 article 6 of the International Convention on the Elimination of All Forms of Racial Discrimination, article 14 of the Convention against Torture and Other Cruel, Inhuman or Degrading Treatment or Punishment, and article 39 of the Convention on the Rights of the Child, and of international humanitarian law as found in article 3 of the Hague Convention respecting the Laws and Customs of War on Land of 18 October 1907 (Convention IV), article 91

¹⁶ Adopted by UN General Assembly Resolution 60/147 (16 December 2005).

of the Protocol Additional to the Geneva Conventions of 12 August 1949, and relating to the Protection of Victims of International Armed Conflicts (Protocol I) of 8 June 1977, and articles 68 and 75 of the Rome Statute of the International Criminal Court,

Recalling the provisions providing a right to a remedy for victims of violations of international human rights found in regional conventions, in particular article 7 of the African Charter on Human and Peoples' Rights, article 25 of the American Convention on Human Rights, and article 13 of the Convention for the Protection of Human Rights and Fundamental Freedoms,

Recalling the Declaration of Basic Principles of Justice for Victims of Crime and Abuse of Power emanating from the deliberations of the Seventh United Nations Congress on the Prevention of Crime and the Treatment of Offenders and General Assembly resolution 40/34 of 29 November 1985 by which the Assembly adopted the text recommended by the Congress,

Reaffirming the principles enunciated in the Declaration of Basic Principles of Justice for Victims of Crime and Abuse of Power, including that victims should be treated with compassion and respect for their dignity, have their right to access to justice and redress mechanisms fully respected, and that the establishment, strengthening and expansion of national funds for compensation to victims should be encouraged, together with the expeditious development of appropriate rights and remedies for victims,

Noting that the Rome Statute of the International Criminal Court requires the establishment of "principles relating to reparations to, or in respect of, victims, including restitution, compensation and rehabilitation", requires the Assembly of States Parties to establish a trust fund for the benefit of victims of crimes within the jurisdiction of the Court, and of the families of such victims, and mandates the Court "to protect the safety, physical and psychological well-being, dignity and privacy of victims" and to permit the participation of victims at all "stages of the proceedings determined to be appropriate by the Court",

Affirming that the Basic Principles and Guidelines contained herein are directed at gross violations

of international human rights law and serious violations of international humanitarian law which, by their very grave nature, constitute an affront to human dignity,

Emphasizing that the Basic Principles and Guidelines contained herein do not entail new international or domestic legal obligations but identify mechanisms, modalities, procedures and methods for the implementation of existing legal obligations under international human rights law and international humanitarian law which are complementary though different as to their norms,

Recalling that international law contains the obligation to prosecute perpetrators of certain international crimes in accordance with international obligations of States and the requirements of national law or as provided for in the applicable statutes of international judicial organs, and that the duty to prosecute reinforces the international legal obligations to be carried out in accordance with national legal requirements and procedures and supports the concept of complementarity,

Noting that contemporary forms of victimization, while essentially directed against persons, may nevertheless also be directed against groups of persons who are targeted collectively,

Recognizing that, in honouring the victims' right to benefit from remedies and reparation, the international community keeps faith with the plight of victims, survivors and future human generations and reaffirms the international legal principles of accountability, justice and the rule of law,

Convinced that, in adopting a victim-oriented perspective, the international community affirms its human solidarity with victims of violations of international law, including violations of international human rights law and international humanitarian law, as well as with humanity at large, in accordance with the following Basic Principles and Guidelines,

Adopts the following Basic Principles and Guide-lines:

(i) Obligation to respect, ensure respect for and implement international human rights law and international humanitarian law

1. The obligation to respect, ensure respect for and implement international human rights law and international humanitarian law as provided for under the respective bodies of law emanates from:

(a) Treaties to which a State is a party;

- (b) Customary international law;
- (c) The domestic law of each State.

2. If they have not already done so, States shall, as required under international law, ensure that their domestic law is consistent with their international legal obligations by:

(*a*) Incorporating norms of international human rights law and international humanitarian law into their domestic law, or otherwise implementing them in their domestic legal system;

(*b*) Adopting appropriate and effective legislative and administrative procedures and other appropriate measures that provide fair, effective and prompt access to justice;

(*c*) Making available adequate, effective, prompt and appropriate remedies, including reparation, as defined below;

(*d*) Ensuring that their domestic law provides at least the same level of protection for victims as that required by their international obligations.

(ii) Scope of the obligation

3. The obligation to respect, ensure respect for and implement international human rights law and international humanitarian law as provided for under the respective bodies of law, includes, inter alia, the duty to:

(*a*) Take appropriate legislative and administrative and other appropriate measures to prevent violations;

(*b*) Investigate violations effectively, promptly, thoroughly and impartially and, where appropriate, take action against those allegedly responsible in accordance with domestic and international law;

(c) Provide those who claim to be victims of a human rights or humanitarian law violation with equal and effective access to justice, as described below, irrespective of who may ultimately be the bearer of responsibility for the violation; and

(*d*) Provide effective remedies to victims, including reparation, as described below.

(iii) Gross violations of international human rights law and serious violations of international humanitarian law that constitute crimes under international law

4. In cases of gross violations of international human rights law and serious violations of international humanitarian law constituting crimes under international law, States have the duty to investigate and, if there is sufficient evidence, the duty to submit to prosecution the person allegedly responsible for the violations and, if found guilty, the duty to punish her or him. Moreover, in these cases, States should, in accordance with international law, cooperate with one another and assist international judicial organs competent in the investigation and prosecution of these violations.

5. To that end, where so provided in an applicable treaty or under other international law obligations, States shall incorporate or otherwise implement within their domestic law appropriate provisions for universal jurisdiction. Moreover, where it is so provided for in an applicable treaty or other international legal obligations, States should facilitate extradition or surrender offenders to other States and to appropriate international judicial bodies and provide judicial assistance and other forms of cooperation in the pursuit of international justice, including assistance to, and protection of, victims and witnesses, consistent with international human rights legal standards and subject to international legal requirements such as those relating to the prohibition of torture and other forms of cruel, inhuman or degrading treatment or punishment.

(iv) Statutes of limitations

6. Where so provided for in an applicable treaty or contained in other international legal obligations, statutes of limitations shall not apply to gross violations of international human rights law and serious violations of international humanitarian law which constitute crimes under international law.

7. Domestic statutes of limitations for other types of violations that do not constitute crimes under international law, including those time limitations applicable to civil claims and other procedures, should not be unduly restrictive.

(v) Victims of gross violations of international human rights law and serious violations of international humanitarian law

8. For purposes of the present document, victims are persons who individually or collectively suffered harm, including physical or mental injury, emotional suffering, economic loss or substantial impairment of their fundamental rights, through acts or omissions that constitute gross violations of international human rights law, or serious violations of international humanitarian law. Where appropriate, and in accordance with domestic law, the term "victim" also includes the immediate family or dependants of the direct victim and persons who have suffered harm in intervening to assist victims in distress or to prevent victimization.

9. A person shall be considered a victim regardless of whether the perpetrator of the violation is identified, apprehended, prosecuted, or convicted and regardless of the familial relationship between the perpetrator and the victim.

(vi) Treatment of victims

10. Victims should be treated with humanity and respect for their dignity and human rights, and appropriate measures should be taken to ensure their safety, physical and psychological well-being and privacy, as well as those of their families. The State should ensure that its domestic laws, to the extent possible, provide that a victim who has suffered violence or trauma should benefit from special consideration and care to avoid his or her retraumatization in the course of legal and administrative procedures designed to provide justice and reparation.

(vii) Victims' right to remedies

11. Remedies for gross violations of international human rights law and serious violations of international humanitarian law include the victim's right to the following as provided for under international law:

(a) Equal and effective access to justice;

(*b*) Adequate, effective and prompt reparation for harm suffered;

(c) Access to relevant information concerning violations and reparation mechanisms.

(viii) Access to justice

12. A victim of a gross violation of international human rights law or of a serious violation of international humanitarian law shall have equal access to an effective judicial remedy as provided for under international law. Other remedies available to the victim include access to administrative and other bodies, as well as mechanisms, modalities and proceedings conducted in accordance with domestic law. Obligations arising under international law to secure the right to access justice and fair and impartial proceedings shall be reflected in domestic laws. To that end, States should:

(*a*) Disseminate, through public and private mechanisms, information about all available remedies for gross violations of international human rights law and serious violations of international humanitarian law;

(*b*) Take measures to minimize the inconvenience to victims and their representatives, protect against unlawful interference with their privacy as appropriate and ensure their safety from intimidation and retaliation, as well as that of their families and witnesses, before, during and after judicial, administrative, or other proceedings that affect the interests of victims;

(*c*) Provide proper assistance to victims seeking access to justice;

(*d*) Make available all appropriate legal, diplomatic and consular means to ensure that victims can exercise their rights to remedy for gross violations of international human rights law or serious violations of international humanitarian law.

13. In addition to individual access to justice, States should endeavour to develop procedures to allow groups of victims to present claims for reparation and to receive reparation, as appropriate.

14. An adequate, effective and prompt remedy for gross violations of international human rights law or serious violations of international humanitarian law should include all available and appropriate international processes in which a person may have legal standing and should be without prejudice to any other domestic remedies.

(ix) Reparation for harm suffered

15. Adequate, effective and prompt reparation is intended to promote justice by redressing gross violations of international human rights law or serious violations of international humanitarian law. Reparation should be proportional to the gravity of the violations and the harm suffered. In accordance with its domestic laws and international legal obligations, a State shall provide reparation to victims for acts or omissions which can be attributed to the State and constitute gross violations of international human rights law or serious violations of international humanitarian law. In cases where a person, a legal person, or other entity is found liable for reparation to a victim, such party should provide reparation to the victim or compensate the State if the State has already provided reparation to the victim.

16. States should endeavour to establish national programmes for reparation and other assistance to victims in the event that the parties liable for the harm suffered are unable or unwilling to meet their obligations.

17. States shall, with respect to claims by victims, enforce domestic judgments for reparation against individuals or entities liable for the harm suffered and endeavour to enforce valid foreign legal judgments for reparation in accordance with domestic law and international legal obligations. To that end, States should provide under their domestic laws effective mechanisms for the enforcement of reparation judgments.

18. In accordance with domestic law and international law, and taking account of individual circumstances, victims of gross violations of international humanitarian law should, as appropriate and proportional to the gravity of the violation and the circumstances of each case, be provided with full and effective reparation, as laid out in principles 19 to 23, which include the following forms: restitution, compensation, rehabilitation, satisfaction and guarantees of non-repetition.

19. *Restitution* should, whenever possible, restore the victim to the original situation before the gross violations of international human rights law or serious violations of international humanitarian law occurred. Restitution includes, as appropriate: restoration of liberty, enjoyment of human rights, identity, family life and citizenship, return to one's place of residence, restoration of employment and return of property.

20. *Compensation* should be provided for any economically assessable damage, as appropriate and proportional to the gravity of the violation and the circumstances of each case, resulting from gross violations of international human rights law and serious violations of international humanitarian law, such as:

(a) Physical or mental harm;

(*b*) Lost opportunities, including employment, education and social benefits;

(c) Material damages and loss of earnings, including loss of earning potential;

(d) Moral damage;

(*e*) Costs required for legal or expert assistance, medicine and medical services, and psychological and social services.

21. *Rehabilitation* should include medical and psychological care as well as legal and social services.

22. *Satisfaction* should include, where applicable, any or all of the following:

(*a*) Effective measures aimed at the cessation of continuing violations;

(*b*) Verification of the facts and full and public disclosure of the truth to the extent that such disclosure does not cause further harm or threaten the safety and interests of the victim, the victim's relatives, witnesses, or persons who have intervened to assist the victim or prevent the occurrence of further violations;

(c) The search for the whereabouts of the disappeared, for the identities of the children abducted, and for the bodies of those killed, and assistance in the recovery, identification and reburial of the bodies in accordance with the expressed or presumed wish of the victims, or the cultural practices of the families and communities;

(*d*) An official declaration or a judicial decision restoring the dignity, the reputation and the rights of the victim and of persons closely connected with the victim;

(*e*) Public apology, including acknowledgement of the facts and acceptance of responsibility;

(*f*) Judicial and administrative sanctions against persons liable for the violations;

(g) Commemorations and tributes to the victims;

(*h*) Inclusion of an accurate account of the violations that occurred in international human rights law and international humanitarian law training and in educational material at all levels.

23. *Guarantees of non-repetition* should include, where applicable, any or all of the following measures, which will also contribute to prevention:

(*a*) Ensuring effective civilian control of military and security forces;

(*b*) Ensuring that all civilian and military proceedings abide by international standards of due process, fairness and impartiality;

(*c*) Strengthening the independence of the judiciary;

(*d*) Protecting persons in the legal, medical and health-care professions, the media and other related professions, and human rights defenders;

(e) Providing, on a priority and continued basis, human rights and international humanitarian law education to all sectors of society and training for law enforcement officials as well as military and security forces;

(*f*) Promoting the observance of codes of conduct and ethical norms, in particular international standards, by public servants, including law enforcement, correctional, media, medical, psychological, social service and military personnel, as well as by economic enterprises;

(*g*) Promoting mechanisms for preventing and monitoring social conflicts and their resolution;

(*h*) Reviewing and reforming laws contributing to or allowing gross violations of international human rights law and serious violations of international humanitarian law.

(x) Access to relevant information concerning violations and reparation mechanisms

24. States should develop means of informing the general public and, in particular, victims of gross violations of international human rights law and serious violations of international humanitarian law of the rights and remedies addressed by these Basic Principles and Guidelines and of all available

legal, medical, psychological, social, administrative and all other services to which victims may have a right of access. Moreover, victims and their representatives should be entitled to seek and obtain information on the causes leading to their victimization and on the causes and conditions pertaining to the gross violations of international human rights law and serious violations of international humanitarian law and to learn the truth in regard to these violations.

(xi) Non-discrimination

25. The application and interpretation of these Basic Principles and Guidelines must be consistent with international human rights law and international humanitarian law and be without any discrimination of any kind or on any ground, without exception.

(xii) Non-derogation

26. Nothing in these Basic Principles and Guidelines shall be construed as restricting or derogating from any rights or obligations arising under domestic and international law. In particular, it is understood that the present Basic Principles and Guidelines are without prejudice to the right to a remedy and reparation for victims of all violations of international human rights law and international humanitarian law. It is further understood that these Basic Principles and Guidelines are without prejudice to special rules of international law.

(xiii) Rights of others

27. Nothing in this document is to be construed as derogating from internationally or nationally protected rights of others, in particular the right of an accused person to benefit from applicable standards of due process.

2. PEACE AGREEMENTS

2.1. Cambodia – Agreement on a Comprehensive Political Settlement on the Cambodia Conflict (1991)

The States participating in the Paris Conference on Cambodia, namely Australia, Brunei Darussalam, Cambodia, Canada, the People's Republic of China, the French Republic, the Republic of India, the Republic of Indonesia, Japan, the Lao People's Democratic Republic, Malaysia, the Republic of the Philippines, the Republic of Singapore, the Kingdom of Thailand, the Union of Soviet Socialist Republics, the United Kingdom of Great Britain and Northern Ireland, the United States of America, the Socialist Republic of Viet Nam and the Socialist Federal Republic of Yugoslavia,

In the presence of the Secretary-General of the United Nations,

In order to maintain, preserve and defend the sovereignty, independence, territorial integrity and inviolability, neutrality and national unity of Cambodia,

Desiring to restore and maintain peace in Cambodia, to promote national reconciliation and to ensure the exercise of the right to self-determination of the Cambodian people through free and fair elections,

Convinced that only a comprehensive political settlement to the Cambodia conflict will be just and durable and will contribute to regional and international peace and security,

Welcoming the Framework document of 28 August 1990, which was accepted by the Cambodian Parties in its entirety as the basis for settling the Cambodia conflict, and which was subsequently unanimously endorsed by Security Council resolution 668 (1990) of 20 September 1990 and General Assembly resolution 45/3 of 15 October 1990,

Noting the formation in Jakarta on 10 September 1990 of the Supreme National Council of Cambodia as the unique legitimate body and source of authority in Cambodia in which, throughout the transitional period, national sovereignty and unity are enshrined, and which represents Cambodia externally,

Welcoming the unanimous election, in Beijing on 17 July 1991, of H.R.H. Prince Norodom Sihanouk as the President of the Supreme National Council,

Recognising that an enhanced United Nations role requires the establishment of a United Nations Transitional Authority in Cambodia (UNTAC) with civilian and military components, which will act with full respect for the national sovereignty of Cambodia,

Noting the statements made at the conclusion of the meetings held in Jakarta on 9–10 September

1990, in Paris on 21–23 December 1990, in Pattaya on 24–26 June 1991, in Beijing on 16–17 July 1991, in Pattaya on 26–29 August 1991, and also the meetings held in Jakarta on 4–6 June 1991 and in New York on 19 September 1991,

Welcoming United Nations Security Council resolution 717 (1991) of 16 October 1991 on Cambodia,

Recognising that Cambodia's tragic recent history requires special measures to assure protection of human rights, and the non-return to the policies and practices of the past,

Have agreed as follows:

[...]

PART V – REFUGEES AND DISPLACED PERSONS

Article 19

Upon entry into force of this Agreement, every effort will be made to create in Cambodia political, economic and social conditions conducive to the voluntary return and harmonious integration of Cambodian refugees and displaced persons.

Article 20

Cambodian refugees and displaced persons, located outside Cambodia, shall have the right to return to Cambodia and to live in safety, security and dignity, free from intimidation or coercion of any kind.

The Signatories request the Secretary-General of the United Nations to facilitate the repatriation in safety and dignity of Cambodian refugees and displaced persons, as an integral part of the comprehensive political settlement and under the overall authority of the Special Representative of the Secretary-General, in accordance with the guidelines and principles on the repatriation of refugees and displaced persons as set forth in annex 4.

[...]

ANNEX 4 – REPATRIATION OF CAMBODIAN REFUGEES AND DISPLACED PERSONS

Part I – Introduction

As part of the comprehensive political settlement, every assistance will need to be given to Cambodian refugees and displaced persons as well as to countries of temporary refuge and the country of origin in order to facilitate the voluntary return of all Cambodian refugees and displaced persons in a peaceful and orderly manner. It must also be ensured that there would be no residual problems for the countries of temporary refuge. The country of origin with responsibility towards its own people will accept their return as conditions become conducive.

Part II – Conditions Conducive to the Return of Refugees and Displaced Persons

The task of rebuilding the Cambodian nation will require the harnessing of all its human and natural resources. To this end, the return to the place of their choice of Cambodians from their temporary refuge and elsewhere outside their country of origin will make a major contribution.

Every effort should be made to ensure that the conditions which have led to a large number of Cambodian refugees and displaced persons seeking refuge in other countries should not recur. Nevertheless, some Cambodian refugees and displaced persons will wish and be able to return spontaneously to their homeland.

There must be full respect for the human rights and fundamental freedoms of all Cambodians, including those of the repatriated refugees and displaced persons, in recognition of their entitlement to live in peace and security, free from intimidation and coercion of any kind. These rights would include, inter alia, freedom of movement within Cambodia, the choice of domicile and employment, and the right to property.

In accordance with the comprehensive political settlement, every effort should be made to create concurrently in Cambodia political, economic and social conditions conducive to the return and harmonious integration of the Cambodian refugees and displaced persons.

With a view to ensuring that refugees and displaced persons participate in the elections, mass repatriation should commence and be completed as soon as possible, taking into account all the political, humanitarian, logistical, technical and socio-economic factors involved, and with the cooperation of the SNC. Repatriation of Cambodian refugees and displaced persons should be voluntary and their decision should be taken in full possession of the facts. Choice of destination within Cambodia should be that of the individual. The unity of the family must be preserved.

PART III - OPERATIONAL FACTORS

Consistent with respect for principles of national sovereignty in the countries of temporary refuge and origin, and in close co-operation with the countries of temporary refuge and origin, full access by the Office of the United Nations High Commissioner for Refugees (UNHCR), CRC and other relevant international agencies should be guaranteed to all Cambodian refugees and displaced persons, with a view to the agencies undertaking the census, tracing, medical assistance, food distribution and other activities vital to the discharge of their mandate and operational responsibilities; such access should also be provided in Cambodia to enable the relevant international organisations to carry out their traditional monitoring as well as operational responsibilities.

In the context of the comprehensive political settlement, the Signatories note with satisfaction that the Secretary-General of the United Nations has entrusted UNHCR with the role of leadership and co-ordination among intergovernmental agencies assisting with the repatriation and relief of Cambodian refugees and displaced persons. The Signatories look to all non-governmental organisations to co-ordinate as much as possible their work for the Cambodian refugees and displaced persons with that of UNHCR.

The SNC, the Governments of the countries in which the Cambodian refugees and displaced persons have sought temporary refuge, and the countries which contribute to the repatriation and integration effort will wish to monitor closely and facilitate the repatriation of the returnees. An ad hoc consultative body should be established for a limited term for these purposes. The UNHCR, the ICRC, and other international agencies as appropriate, as well as UNTAC, would be invited to join as full participants.

Adequately monitored short-term repatriation assistance should be provided on an impartial basis to enable the families and individuals returning to Cambodia to establish their lives and livelihoods harmoniously in their society. These interim measures would be phased out and replaced in the longer term by the reconstruction programme.

Those responsible for organising and supervising the repatriation operation will need to ensure that conditions of security are created for the movement of the refugees and displaced persons. In this respect, it is imperative that appropriate border crossing points and routes be designated and cleared of mines and other hazards.

The international community should contribute generously to the financial requirements of the repatriation operation.

[...]

2.2. El Salvador – Peace Agreement between the Government of El Salvador and the Frente Farabundo Martí para la Liberación Nacional (1992)

[...]

CHAPTER V

Economic and Social Questions

1 Preamble

One of the prerequisites for the democratic reunification of Salvadorian society is the sustained economic and social development of the country. At the same time, reunification of Salvadorian society and a growing degree of social cohesion are indispensable for fostering development. Hence, the set of agreements required to put a definitive end to the armed conflict in El Salvador must include certain minimum commitments to promote development for the benefit of all sectors of the population.

In accordance with the New York Agreement, the issues covered by this instrument are: the agrarian problem, loans to the agricultural sector, measures required to alleviate the social cost of structural adjustment programmes, appropriate procedures for direct external cooperation designed to encourage community development and assistance projects, establishment of a forum for economic and social consultation and the National Reconstruction Plan. Also, although the general philosophy or orientation of the Government's economic policy, which FMLN does not necessarily share, is not covered by this Agreement, both Parties agree on the need to provide certain basic guidelines so as to ensure the requisite social stability during the transitional period, consolidate peace and make progress towards the reunification of Salvadorian society.

2 The agrarian problem

Lands in excess of the constitutional limit of 245 hectares

The Government of El Salvador shall transfer rural farmland that has not yet been transferred under articles 105 and 267 of the Constitution of the Republic.

It likewise undertakes to ensure that implementation of the relevant constitutional requirements is not evaded by owners of rural holdings in excess of 245 hectares.

State-owned lands which are not currently part of a forestry reserve

The Government of El Salvador shall transfer to beneficiaries of the agrarian reform, as provided in article 104 of the Constitution, State-owned rural farmland which is not part of a forestry reserve.

Under the various land-transfer programmes which the Government of El Salvador is carrying out with State-owned farmland, preference shall be given to former combatants of both Parties who so request voluntarily, are of peasant origin and familiar with farming, and possess no land of any kind. The size of the lots shall be determined by the amount of land available, as mentioned above, and the number of beneficiaries who meet the conditions set out in this section.

Lands offered for sale to the State

Making use of the legal, technical and financial resources available to it, the Government of El Salvador shall seek to acquire and transfer through the Land Bank lands voluntarily offered for sale by their owners. Once the said lands are acquired, they shall be transferred to beneficiaries of the agrarian reform.

Recipients of lands transferred in accordance with the preceding sections

The lands acquired under sections A, B and C of this chapter shall be used to satisfy the need for land of landless peasants and small farmers. Specifically, title to the land shall be transferred legally to the peasants and small farmers designated by law as beneficiaries of the agrarian reform.

Payments for land

The lands referred to in the preceding sections shall be transferred at market prices and on the same credit terms as are granted to beneficiaries of the reformed sector. At the same time, a system of payments may be established on the basis of a fixed price and long-term financing at low, fixed interest rates not subject to interest capitalization. Domestic credit shall be supplemented with financing from international cooperation, for which a special fund, financed from external resources, shall be established for the purchase of land.

New legislation

Since the current agrarian legislation is haphazard, contradictory and incomplete, the Parties agree that it must be harmonized and unified into an agrarian code. To this end, the Government shall submit the relevant draft legislation to the Legislative Assembly no later than 12 months after the signing of this Agreement. If it fails to do so, COPAZ shall take on the task of preparing the corresponding preliminary draft.

3 Lands within conflict zones

The land-tenure system in conflict zones

In accordance with the New York Agreement, the current land-tenure situation in conflict zones shall be respected until a satisfactory legal solution for the definitive land-tenure system is arrived at. Consequently, landholders shall not be evicted pending agreement on such a solution; moreover, they shall be given financial support to increase agricultural production.

In view of the irregularity of the land-tenure system in conflict zones, the Parties agree on the following:

Determination as to who are the "current landholders"

"Landholders" shall mean those currently occupying and/or working the land in conflict zones.

Inventory of cases covered by this part of the Agreement

Within 30 days from the signing of the Agreement, FMLN shall submit an inventory of land or buildings affected by the Agreement. Upon verification that such land or buildings are in fact subject to the provisions of this Agreement, and in accordance with the procedure set forth in the next section, the Government of El Salvador shall seek to provide a satisfactory legal solution for their final disposal through the voluntary sale of such property by the rightful owners to the current holders, on the terms referred to in section 3(F) of this chapter.

Should a rightful owner not wish to sell his property, the Government of El Salvador shall make use of the legal mechanisms at its disposal to try to resettle the peasants or small farmers on such land as may be available for the purpose and shall, as far as possible, seek to ensure that such land is situated in the same zones.

Establishment of a Special Commission

COPAZ shall appoint a special commission whose members shall be of recognized integrity and ability. The special commission, to be formed within 20 days following the signing of this Agreement, shall be entrusted with the following tasks and duties:

To verify the inventory of affected land or buildings within conflict zones. Once the inventory has been verified, the special commission shall submit copies to the Government of El Salvador and to COPAZ;

Should the need arise, to facilitate the settlement of disputes between current holders and rightful owners;

To take any decisions and measures it deems necessary and proper for the prompt and effective fulfilment of the agreements set forth in this chapter.

Legalization of land tenure

Except for particularly complex cases, the Government of El Salvador shall legalize the land-tenure situation in conflict zones definitively within six months from the signing of the cease-fire agreement, granting, as appropriate, individual or collective title to the land.

Payment for lands

Lands shall be purchased from their former owners at market prices. The sale to the current holders shall be subject to the same conditions as those granted to beneficiaries of the reformed sector. However, special conditions may be agreed to in the interests of the peace process.

Verification by COPAZ

COPAZ shall guarantee fulfilment of the agreements set forth in sections 2 and 3.

4 3 July 1991 agreement on occupied lands

The agreement on occupied lands between the Government of El Salvador and peasant organizations shall be respected.

With regard to lands occupied illegally after the date of that agreement, the Government of El Salvador gives notice that it reserves the right to enforce the relevant legal provisions so as to ensure that the rule of law prevails. FMLN holds that the agrarian problem, including land occupations, should be dealt with through consultation and the channels and mechanisms provided by the peace agreements.

[...]

2.3. Mozambique – The Rome Process: General Peace Agreement for Mozambique (1992)

[...]

PROTOCOL III

(iv) Return of Mozambican refugees and displaced persons and their social reintegration

(a) The parties undertake to co-operate in the repatriation and reintegration of Mozambican refugees and displaced persons in the national territory and the social integration of war-disabled;

(b) Without prejudice to the liberty of movement of citizens, the Government shall draw up a draft agreement with Renamo to organise the necessary assistance to refugees and displaced persons, preferably in their original places of residence. The parties agree to seek the involvement of the competent United Nations agencies in the drawing up and implementation of this plan. The International Red Cross and other organisations to be agreed upon shall be invited to participate in the implementation of the plan;

(c) Mozambican refugees and displaced persons shall not forfeit any of the rights and freedoms of citizens for having left their original places of residence;

(d) Mozambican refugees and displaced persons shall be registered and included in the electoral rolls together with other citizens in their places of residence;

(e) Mozambican refugees and displaced persons shall be guaranteed restitution of property owned by them which is still in existence and the right to take legal action to secure the return of such property from individuals in possession of it.

[...]

2.4. Liberia - Cotonou Agreement (1993)

[...]

Article 18 - Repatriation of Refugees

The Parties hereby commit themselves immediately and permanently to bring to an end any further external or internal displacement of Liberians and to create the conditions that will allow all refugees and displaced persons to, respectively, voluntarily repatriate and return to Liberia to their places of origin or habitual residence under conditions of safety and dignity.

The Parties further call upon Liberian refugees and displaced persons to return to Liberia and to their places of origin or habitual residence and declare that they shall not be jeopardized in any ethnic, political, religious, regional or geographical considerations.

The Parties also call upon the relevant organizations of the United Nations system, particularly the Office of the United Nations High Commissioner for Refugees and the United Nations Development Programme, other intergovernmental and nongovernmental organizations, to implement programmes for the voluntary repatriation, return and reintegration of the Liberian refugees and internally displaced persons.

The Parties proclaim that they shall, jointly or individually, cooperate in all necessary ways with themselves and with the above-mentioned organizations in order to facilitate the repatriation, return and reintegration of the refugees and displaced persons. Amongst others, they agree to:

Establish all necessary mechanisms or arrangements, such as joint repatriation committees, which would facilitate contacts, communications and work with the relevant organizations for purposes of implementing the repatriation, return and reintegration operation and to enable effective decision-making and implementation of the relevant activities;

Facilitate access by the Office of the United Nations High Commissioner for Refugees and other organizations to the refugees and displaced persons who have returned so as to deliver the necessary humanitarian assistance and programmes and monitor their situation;

Guarantee and provide security to the Office of the United Nations High Commissioner for Refugees and the other relevant organizations, their staff, vehicles, equipment and resources necessary to carry out their work;

Provide all other necessary facilities and support that will be necessary to facilitate the implementation of the return, voluntary repatriation and reintegration of refugees and displaced persons.

[...]

2.5. Rwanda - Arusha Peace Agreement (1993)

[...]

Article 4

The right to property is a fundamental right for all the people of Rwanda. All refugees shall therefore have the right to repossess their property on return.

The two parties recommend, however, that in order to promote social harmony and national reconciliation, refugees who left the country more than 10 years ago should not reclaim their properties, which might have been occupied by other people. The Government shall compensate them by putting land at their disposal and shall help them to resettle. As for estates which have been occupied by the Government, the returnee shall have the right for an equitable compensation by the Government.

[...]

2.6. Bosnia-Herzegovina – General Framework Agreement on Peace in Bosnia-Herzegovina (Dayton Peace Agreement) – Annex 7: Agreement on Refugees and Displaced Persons (1995)

The Republic of Bosnia and Herzegovina, the Federation of Bosnia and Herzegovina, and the Republika Srpska (the "Parties") have agreed as follows:

CHAPTER ONE: PROTECTION

Article I – Rights of Refugees and Displaced Persons

1. All refugees and displaced persons have the right freely to return to their homes of origin. They shall have the right to have restored to them property of which they were deprived in the course of hostilities since 1991 and to be compensated for any property that cannot be restored to them. The early return of refugees and displaced persons is an important objective of the settlement of the conflict in Bosnia and Herzegovina. The Parties confirm that they will accept the return of such persons who have left their territory, including those who have been accorded temporary protection by third countries.

2. The Parties shall ensure that refugees and displaced persons are permitted to return in safety, without risk of harassment, intimidation, persecution, or discrimination, particularly on account of their ethnic origin, religious belief, or political opinion.

3. The Parties shall take all necessary steps to prevent activities within their territories which would hinder or impede the safe and voluntary return of refugees and displaced persons. To demonstrate their commitment to securing full respect for the human rights and fundamental freedoms of all persons within their jurisdiction and creating without delay conditions suitable for return of refugees and displaced persons, the Parties shall take immediately the following confidence building measures:

(a) the repeal of domestic legislation and administrative practices with discriminatory intent or effect;

(b) the prevention and prompt suppression of any written or verbal incitement, through media or otherwise, of ethnic or religious hostility or hatred;

(c) the dissemination, through the media, of warnings against, and the prompt suppression of, acts of retribution by military, paramilitary, and police services, and by other public officials or private individuals;

(d) the protection of ethnic and/or minority populations wherever they are found and the provision of immediate access to these populations by international humanitarian organizations and monitors;

(e) the prosecution, dismissal or transfer, as appropriate, of persons in military, paramilitary, and police forces, and other public servants, responsible for serious violations of the basic rights of persons belonging to ethnic or minority groups.

4. Choice of destination shall be up to the individual or family, and the principle of the unity of the family shall be preserved. The Parties shall not interfere with the returnees' choice of destination, nor shall they compel them to remain in or move to situations of serious danger or insecurity, or to areas lacking in the basic infrastructure necessary to resume a normal life. The Parties shall facilitate the flow of information necessary for refugees and displaced persons to make informed judgments about local conditions for return.

5. The Parties call upon the United Nations High Commissioner for Refugees ("UNHCR") to develop in close consultation with asylum countries and the Parties a repatriation plan that will allow for an early, peaceful, orderly and phased return of refugees and displaced persons, which may include priorities for certain areas and certain categories of returnees. The Parties agree to implement such a plan and to conform their international agreements and internal laws to it. They accordingly call upon States that have accepted refugees to promote the early return of refugees consistent with international law.

Article II – Creation of Suitable Conditions for Return

1. The Parties undertake to create in their territories the political, economic, and social conditions conducive to the voluntary return and harmonious reintegration of refugees and displaced persons, without preference for any particular group. The Parties shall provide all possible assistance to refugees and displaced persons and work to facilitate their voluntary return in a peaceful, orderly and phased manner, in accordance with the UNHCR repatriation plan.

2. The Parties shall not discriminate against returning refugees and displaced persons with respect to conscription into military service, and shall give positive consideration to requests for exemption from military or other obligatory service based on individual circumstances, so as to enable returnees to rebuild their lives.

Article III – Cooperation with International Organizations and International Monitoring

1. The Parties note with satisfaction the leading humanitarian role of UNHCR, which has been entrusted by the Secretary-General of the United Nations with the role of coordinating among all agencies assisting with the repatriation and relief of refugees and displaced persons.

2. The Parties shall give full and unrestricted access by UNHCR, the International Committee of the Red Cross ("ICRC"), the United Nations Development Programme ("UNDP"), and other relevant international, domestic and nongovernmental organizations to all refugees and displaced persons, with a view to facilitating the work of those organizations in tracing persons, the provision of medical assistance, food distribution, reintegration assistance, the provision of temporary and permanent housing, and other activities vital to the discharge of their mandates and operational responsibilities without administrative impediments. These activities shall include traditional protection functions and the monitoring of basic human rights and humanitarian conditions, as well as the implementation of the provisions of this Chapter.

3. The Parties shall provide for the security of all personnel of such organizations.

Article IV - Repatriation Assistance

The Parties shall facilitate the provision of adequately monitored, short-term repatriation assistance on a non-discriminatory basis to all returning refugees and displaced persons who are in need, in accordance with a plan developed by UNHCR and other relevant organizations, to enable the families and individuals returning to re-establish their lives and livelihoods in local communities.

Article V - Persons Unaccounted For

The Parties shall provide information through the tracing mechanisms of the ICRC on all persons unaccounted for. The Parties shall also cooperate fully with the ICRC in its efforts to determine the identities, whereabouts and fate of the unaccounted for.

Article VI - Amnesty

Any returning refugee or displaced person charged with a crime, other than a serious violation of international humanitarian law as defined in the Statute of the International Tribunal for the Former Yugoslavia since January 1, 1991 or a common crime unrelated to the conflict, shall upon return enjoy an amnesty. In no case shall charges for crimes be imposed for political or other inappropriate reasons or to circumvent the application of the amnesty.

CHAPTER TWO: COMMISSION FOR DISPLACED Persons and Refugees

Article VII - Establishment of the Commission

The Parties hereby establish an independent Commission for Displaced Persons and Refugees (the "Commission"). The Commission shall have its headquarters in Sarajevo and may have offices at other locations as it deems appropriate.

Article VIII - Cooperation

The Parties shall cooperate with the work of the Commission, and shall respect and implement its decisions expeditiously and in good faith, in cooperation with relevant international and nongovernmental organizations having responsibility for the return and reintegration of refugees and displaced persons.

Article XI – Mandate

The Commission shall receive and decide any claims for real property in Bosnia and Herzegovina, where the property has not voluntarily been sold or otherwise transferred since April 1, 1992, and where the claimant does not now enjoy possession of that property. Claims may be for return of the property or for just compensation in lieu of return.

Article XII - Proceedings before the Commission

1. Upon receipt of a claim, the Commission shall determine the lawful owner of the property with respect to which the claim is made and the value of that property. The Commission, through its staff or a duly designated international or nongovernmental organization, shall be entitled to have access to any and all property records in Bosnia and Herzegovina, and to any and all real property located in Bosnia and Herzegovina for purposes of inspection, evaluation and assessment related to consideration of a claim.

2. Any person requesting the return of property who is found by the Commission to be the lawful owner of that property shall be awarded its return. Any person requesting compensation in lieu of return who is found by the Commission to be the lawful owner of that property shall be awarded just compensation as determined by the Commission. The Commission shall make decisions by a majority of its members.

3. In determining the lawful owner of any property, the Commission shall not recognize as valid any illegal property transaction, including any transfer that was made under duress, in exchange for exit permission or documents, or that was otherwise in connection with ethnic cleansing. Any person who is awarded return of property may accept a satisfactory lease arrangement rather than retake possession.

4. The Commission shall establish fixed rates that may be applied to determine the value of all real property in Bosnia and Herzegovina that is the subject of a claim before the Commission. The rates shall be based on an assessment or survey of properties in the territory of Bosnia and Herzegovina undertaken prior to April 1, 1992, if available, or may be based on other reasonable criteria as determined by the Commission. 5. The Commission shall have the power to effect any transactions necessary to transfer or assign title, mortgage, lease, or otherwise dispose of property with respect to which a claim is made, or which is determined to be abandoned. In particular, the Commission may lawfully sell, mortgage, or lease real property to any resident or citizen of Bosnia and Herzegovina, or to either Party, where the lawful owner has sought and received compensation in lieu of return, or where the property is determined to be abandoned in accordance with local law. The Commission may also lease property pending consideration and final determination of ownership.

6. In cases in which the claimant is awarded compensation in lieu of return of the property, the Commission may award a monetary grant or a compensation bond for the future purchase of real property. The Parties welcome the willingness of the international community assisting in the construction and financing of housing in Bosnia and Herzegovina to accept compensation bonds awarded by the Commission as payment, and to award persons holding such compensation bonds priority in obtaining that housing.

7. Commission decisions shall be final, and any title, deed, mortgage, or other legal instrument created or awarded by the Commission shall be recognized as lawful throughout Bosnia and Herzegovina.

8. Failure of any Party or individual to cooperate with the Commission shall not prevent the Commission from making its decision.

Article XIII - Use of Vacant Property

The Parties, after notification to the Commission and in coordination with UNHCR and other international and nongovernmental organizations contributing to relief and reconstruction, may temporarily house refugees and displaced persons in vacant property, subject to final determination of ownership by the Commission and to such temporary lease provisions as it may require.

Article XIV – Refugees and Displaced Persons Property Fund

1. A Refugees and Displaced Persons Property Fund (the "Fund") shall be established in the Central Bank of Bosnia and Herzegovina to be administered by the Commission. The Fund shall be replenished through the purchase, sale, lease and mortgage of real property which is the subject of claims before the Commission. It may also be replenished by direct payments from the Parties, or from contributions by States or international or nongovernmental organizations.

2. Compensation bonds issued pursuant to Article XII(6) shall create future liabilities on the Fund under terms and conditions to be defined by the Commission.

Article XV - Rules and Regulations

The Commission shall promulgate such rules and regulations, consistent with this Agreement, as may be necessary to carry out its functions. In developing these rules and regulations, the Commission shall consider domestic laws on property rights.

Article XVI – Transfer

Five years after this Agreement takes effect, responsibility for the financing and operation of the Commission shall transfer from the Parties to the Government of Bosnia and Herzegovina, unless the Parties otherwise agree. In the latter case, the Commission shall continue to operate as provided above.

Article XVII - Notice

The Parties shall give effective notice of the terms of this Agreement throughout Bosnia and Herzegovina, and in all countries known to have persons who were citizens or residents of Bosnia and Herzegovina.

[...]

2.7. Guatemala – Agreement on Identity and Rights of Indigenous Peoples (1995)

[...]

F. Rights Relating to Land of the Indigenous Peoples

1. The rights relating to land of the indigenous peoples include both the communal or collective and the individual tenure of land, rights of ownership and possession and other real rights, and the use of natural resources for the benefit of the communities without detriment to their habitat. Legislative and administrative measures must be developed to ensure recognition, the awarding of title, protection, recovery, restitution and compensation for those rights.

2. The lack of protection of the rights relating to land and natural resources of the indigenous peoples is part of a very wide-ranging set of problems resulting, inter alia, from the fact that both the indigenous and the non-indigenous peasants have had difficulty in having their rights legalized through the acquisition of title and land registration. When, in exceptional cases, they have been able to have their rights legalized, they have not had access to legal mechanisms to defend them. Since this problem is not exclusive to the indigenous population - although the latter has been particularly affected - it should be dealt with in the context of "Social and economic issues and the agrarian question", as one of the considerations to be taken into account in connection with the reform of the land tenure structure.

3. However, the situation with regard to the particular lack of protection and plundering of indigenous communal or collectively held lands merits special attention within the framework of this agreement. The Guatemalan Constitution establishes the obligation of the State to give special protection to cooperative, communal or collectivelyheld lands; recognizes the right of indigenous and other communities to maintain the system of administration of the lands which they hold and which historically belong to them; and lays down the obligation of the State to provide State lands for the indigenous communities which need them for their development.

4. Recognizing the special importance which their relationship to the land has for the indigenous communities, and in order to strengthen the exercise of their collective rights to the land and its natural resources, the Government undertakes to adopt directly, when that is within its competence, and to promote, when that is within the competence of the legislative organ or the municipal authorities, the following measures, inter alia, which shall be implemented in consultation and coordination with the indigenous communities concerned.

Regularization of the land tenure of indigenous communities

5. The Government shall adopt or promote measures to regularize the legal situation with regard to the communal possession of lands by communities which do not have the title deeds to those lands, including measures to award title to municipal or national lands with a clear communal tradition. To that end, an inventory of the land tenure situation shall be drawn up in each municipality.

Land tenure and use and administration of natural resources

6. The Government shall adopt or promote the following measures:

(a) Recognize and guarantee the right of access to lands and resources which are not occupied exclusively by communities but to which the latter have historically had access for their traditional activities and their subsistence (rights of way, such as passage, wood-cutting, access to springs, etc., and use of natural resources) and for their spiritual activities;

(b) Recognize and guarantee the right of communities to participate in the use, administration and conservation of the natural resources existing in their lands;

(c) Secure the approval of the indigenous communities prior to the implementation of any project for the exploitation of natural resources which might affect the subsistence and way of life of the communities. The communities affected shall receive fair compensation for any loss which they may suffer as a result of these activities; and

(d) Adopt, in cooperation with the communities, the measures necessary for the protection and preservation of the environment.

Restitution of communal lands and compensation for rights

7. Recognizing the particularly vulnerable situation of the indigenous communities, which have historically been the victims of land plundering, the Government undertakes to institute proceedings to settle the claims to communal lands formulated by the communities and to restore or pay compensation for those lands. In particular, the Government shall adopt or promote the following measures:

(a) Suspend the awarding of supplementary titles in respect of property to which the indigenous communities have claimed a right;

(b) Suspend the statute of limitations in respect of any action involving the plundering of the indigenous communities; and

(c) When the statute of limitations has already expired, however, establish procedures to compensate the communities which have been plundered with lands acquired for that purpose.

Acquisition of land for the development of indigenous communities

8. The Government shall take the necessary measures, without detriment to peasant smallholdings, to discharge its constitutional mandate to provide State lands for the indigenous communities which need them for their development.

Legal protection of the rights of indigenous communities

9. In order to facilitate the defence of the aforementioned rights and to protect the communities effectively, the Government undertakes to adopt or promote the following measures:

(a) Develop legal rules recognizing the right of indigenous communities to administer their lands in accordance with their customary norms;

(b) Promote an increase in the number of courts dealing with land cases and expedite procedures for the settlement of those cases;

(c) Urge faculties of law and the social sciences to strengthen the agrarian law component of the curriculum and include a knowledge of the relevant customary norms;

(d) Establish competent legal advisory services to advise on land claims;

(e) Provide the indigenous communities with the services of interpreters, free of charge, in respect of legal matters;

(f) Promote the widest dissemination, within indigenous communities, of information about land rights and the legal recourses available; and

(g) Eliminate any form of discrimination against women, in fact or in law, with regard to facilitating access to land, housing, loans and participation in development projects.

10. The Government undertakes to give the fulfilment of the undertakings set out in this section F the priority which the situation of insecurity and urgency that characterize the land problems of the indigenous communities deserves. To that end, the Government shall, in consultation with the indigenous peoples, establish a joint commission on the rights relating to land of the indigenous peoples to study, devise and propose more appropriate institutional arrangements and procedures. The commission shall be composed of representatives of the Government and of indigenous organizations.

[...]

2.8. Croatia - The Erdut Agreement (1995)

BASIC AGREEMENT ON THE REGION OF EASTERN Slavonia, Baranja, and Western Sirmium

The Parties agree as follows:

1. There shall be a transitional period of 12 months which may be extended at most to another period of the same duration if so requested by one of the parties.

2. The U.N. Security Council is requested to establish a Transitional Administration, which shall govern the Region during the transitional period in the interest of all persons resident in or returning to the Region.

3. The U.N. Security Council is requested to authorize an international force to deploy during the transitional period to maintain peace and security in the Region and otherwise to assist in implementation of this Agreement. The Region shall be demilitarized according to the schedule and procedures determined by the international force. This demilitarization shall be completed not later than 30 days after deployment of the international force and shall include all military forces, weapons, and police, except for the international force and for police operating under the supervision of, or with the consent of, the Transitional Administration.

4. The Transitional Administration shall ensure the possibility for the return of refugees and displaced persons to their homes of origin. All persons who have left the Region or who have come to the Region with previous permanent residence in Croatia shall enjoy the same rights as all other residents of the Region. The Transitional Administration shall also take the steps necessary to re-establish the normal functioning of all public services in the Region without delay.

5. The Transitional Administration shall help to establish and train temporary police forces, to build